# **FINAL REPORT**

### **TNO-report**

R 97/435

Development and demonstration of a method to monitor the effects of measures to reduce VOC emissions in the EU. LIFE 95/NL/A11/NL/365/ZHL [1966-1998] (Final Report) Laan van Westenenk 501 Postbus 342 7300 AH Apeldoorn The Netherlands

www.mep.tno.nl

T +31 55 549 34 93 F +31 55 541 98 37 info@mep.tno.nl

Date	January 1998
Authors	Dr. M.P. Keuken Dr. R.F. van Oss
Order no.	53397
Keywords	<ul> <li>VOC monitoring strategy</li> <li>modelling</li> <li>industrial area</li> <li>Rijnmond-the Netherlands</li> </ul>
Intended for	Mr. C. Pleinveaux, European Commission DG XI.B2

All rights reserved.

No part of this publication may be reproduced and/or published by print, photoprint, microfilm or any other means without the previous written consent of TNO.

In case this report was drafted on instructions, the rights and obligations of contracting parties are subject to either the Standard Conditions for Research Instructions given to TNO, or the relevant agreement concluded between the contracting parties.

Submitting the report for inspection to parties who have a direct interest is permitted.

© 2002 TNO

TNO-MEP - R 97/435

2 van 87

Project Characteristics	Description						
Name	Development and demonstration of a method to monitor the effects of measures to reduce VOC emissions in the EU						
Location	Rijnmond area, Rotterdam, the Netherlands						
Contractor	TNO, the Netherlands						
Duration	1/1/1996 - 1/2/1998						
Objectives	To develop and demonstrate a methodology to assess VOC emissions, and to estimate the contributions by various source categories in a complex industrial area based upon ground level VOC measurements.						
Activities	Phase I: 1/1/1996-1/4/1997						
	<ol> <li>to develop a modelling methodology;</li> </ol>						
	<ol><li>to select a measuring strategy; and</li></ol>						
	<ol><li>to consult national and international experts.</li></ol>						
	Phase II: 1/4/1997-1/2/1998						
	1) to implement the measuring strategy;						
	<ol> <li>to process the data in accordance with the modelling methodology; and</li> </ol>						
	<ol><li>to report the outcome of the project.</li></ol>						
Results	<ul> <li>demonstration of the methodology at Rijnmond, Rotterdam;</li> <li>description and dissemination of the methodology.</li> </ul>						
Project costs *1000	ECU 430,- (NLG. 897,-)						
Life contribution * 1000	ECU 209,- (NLG. 436,-)						
	NLG. 375,- staffing;						
	NLG. 15,- travelling;						
	NLG. 18,- equipment;						
	NLG. 14,- operational;						
	NLG. 8,- dissemination/reporting;						
NLG. 6,- contingencies;							
VROM contribution * 1000	NLG. 250,-						
SHELL contribution * 1000	00 NLG. 100,- (in kind)						
DCMR contribution * 1000	NLG. 100,- (in kind)						

3 of 87

TNO-MEP - R 97/435

### **Executive summary**

In 1996, TNO was granted by the European Commission to carry out in the framework of the LIFE programme a project entitled "Development and demonstration of a method to monitor the effects of measures to reduce VOC emissions in the EU". An additional grant was supplied by the Dutch Ministry for Housing, Physical Planning and the Environment. Two other organisations also contributed to the project: Shell Research Ltd., the Netherlands and the Environmental Protection Agency of Rijnmond (DCMR).

#### background

Ambient air concentrations of Volatile Organic Compounds (VOC) may cause health effects either directly by toxic compounds such as benzene and/or indirectly by formation of ground-level ozone. Next to traffic, industrial activities (i.e. chemical base, storage tanks and refineries) are important sources of VOC emissions. In order to implement cost-effective emission-reduction measures and to monitor the impact of these measures, information is required on stack and diffuse emissions of industrial areas.

#### project goal

In the Netherlands, the National Emission Inventory (EI) is used to collect information on VOC emissions. Subsequently, in combination with dispersion models, ground-level air concentrations of VOC are computed around industrial areas. The main constraints to apply this methodology are *firstly*, discrepancies with measurements of ground-level air concentrations, *secondly*; air concentrations estimated from EI data concern *annual* average concentrations and hence, there is considerable delay to provide this information and no information is available on fluctuations within a year for example during a smogepisode, and *thirdly*; EI data do not exist in all European countries. In order to address these constraints, *the goal of the project is to develop an alternative method to assess VOC emissions from industrial areas*.

#### project objectives

The objectives of the project are: i.) to develop a methodology to assess VOC emissions from complex industrial areas based upon VOC ground-level measurements and ii.) to demonstrate the methodology in the Rijnmond area by identification of various sources of VOC emissions and computation of the strength of the emission rates.

#### project methodology

After development of the measuring and modelling strategy and selection of a representative industrial area within the Rijnmond region, a measuring campaign has been implemented for a three month period. Subsequently, the measurement results have been used for model calculations of the emission rates and

TNO-MEP - R 97/435

6 van 87

compared with values of the National Emission Inventory. Finally, a method is recommended to assess VOC emissions from complex industrial areas.

In order to develop a methodology, two types of ground-level measurements of VOC have been tested: within and outside an industrial area. For " $C_6 - C_8$ " hydrocarbons, passive samplers with a sampling period of two weeks have been applied, while for " $C_2$ - $C_5$ " and " $C_6$ - $C_8$ " hydrocarbons hourly on-line measurements and active sampling have been performed. Passive sampling is cost-effective and provides information especially on *spatial* variation of ground-level air concentrations within an industrial area. On-line measurements and active sampling are more costly and provides information especially on *temporal* variation of ground-level air concentrations up- and down-wind outside an industrial area. In addition to these rather traditional measurement concepts, also remote sensing methods were applied.

In the project, the following so-called " $C_2$ - $C_8$ " hydrocarbons have been measured: ethane, ethene, acetylene, propane, n-butane, i-butane, n-pentane, i-pentane, total hexane, benzene, total heptane, toluene, total octane, o,m,p-xylene, ethylbenzene and styrene. For practical reasons, model calculations have been limited to three components: benzene (" $C_6$ "), styrene (" $C_8$ ") and ethene (" $C_2$ "). These compounds were selected for various reasons: benzene for the concern about health effects and the combination of low and stack emission sources, while styrene is only emitted by a limited number of stack emission sources and ethene as a representative of " $C_2$ - $C_5$ " hydrocarbons, which are important precursors for ground-level ozone.

#### project activities

The project period was from 1 January 1996 till 1 February 1998. The project consists of two phases: Phase I (1/1/96-1/4/97) to develop a measuring and modelling strategy, and to select a demonstration area within the Rijnmond region, and Phase II (1/4/97-1/2/98) to demonstrate the methodology, and to report and disseminate the results of the project. The project has been completed within the budget of the project and envisaged staffing inputs.

#### results of phase I of the project

A modelling methodology; The developed methodology in phase I is based upon a combination of factor analysis, spatial linear regression, Bayesian statistics and the updated, short-term National Plume model. The method involves a number of steps: 1.) correction of data for background and traffic emissions, 2.) presentation of hourly measured air concentrations in accordance to wind sectors, 3.) factor analysis to identify source categories of VOC emissions, and 4.) spatial linear regression, Bayesian statistics and inverse-dispersion modelling to compute emission rates.

A measuring strategy; The developed strategy involves a triangle of three monitoring stations *outside* the industrial area to measure hourly " $C_2$ - $C_5$ " concentrations. This approach provides ground-level concentration differences at six wind directions in which one station serves as receptor station (down-wind of the industrial area), while the opposite station is the background station (up-wind of the industrial area). Fourteen passive sampling locations *within* and three locations *outside* the industrial area were applied to measure " $C_6$ - $C_8$ " concentrations. The number of locations within the industrial area was related to (expected) number of emission sources, while outside the area the locations were similar to the " $C_2$ - $C_5$ " monitoring stations.

A demonstration area; Within the Rijnmond region, the Botlek/Pernis industrial area was selected as demonstration with a surface of 4\*2 km. Industrial activities within the Botlek/Pernis area include large-scale chemical plants, refineries and storage of oil products.

#### results of phase II of the project

*Results*; Taking into account uncertainties of meteorological and air concentrations measurements, as well as comparing three-monthly measurements in 1997 with *annual* Emission Inventory data from 1995, the analysis of benzene, styrene and ethene demonstrate that both passive and active sampling result in comparable emission rates within a factor two of the Emission Inventory data.

Data analysis of the measuring campaign indicate that the *styrene* emission rate of the most important source in Botlek/Pernis area is a factor 10 smaller compared to the EI database of 1995 (according to information from Shell, the result of recent measures to reduce styrene emissions); *benzene* emission rates of the 27 most important sources agree within a factor 2-4 with the EI database, with the exception of stack emission at 213 m height where the mixing height may influence the method and an unknown *ethene* emission source was identified. The measured two-weekly averaged emission rates of styrene and benzene show fluctuations over an order of magnitude and consequently, also ground-level air concentrations of these compounds.

#### conclusions

It is concluded that the results presented in this report for styrene, benzene and ethene demonstrate that the method provides adequate information on the *spatial distribution* of emission sources, *chemical profiles* of emission sources, *temporal fluctuations* of emission rates and the *reliability* of the calculated emission rates. The method can distinguish low (diffuse) and high (stack) emissions.

8 van 87

A second conclusion is that the most cost-effective and simple methodology to assess emission rates from low sources of "C<sub>6</sub> and higher VOC" (including chlorinated compounds) is based upon passive sampling *within* the industrial area. For "C<sub>2</sub>-C<sub>5</sub> VOC" emissions, a triangle of up- and downwind receptor locations are required based upon hourly on-line measurements and/or active sampling.

A third conclusion is that the developed method provides an important instrument for enforcement agencies to monitor actual VOC emissions from a complex industrial sources, not only on an annual but even at a two-weekly basis. This is an important aspect in order to check the effects of measures to reduce industrial VOC emissions.

A fourth conclusion concerns VOC emission measurements by the HAWK and FTIR remote sensing techniques. It is concluded that the HAWK is not an adequate technique to estimate VOC emissions from an industrial area, due to its lack of sensitivity for " $C_2$ - $C_5$  VOC", as compared to methane air concentrations. Under plant conditions with a more favourable " $C_2$ - $C_5$  VOC"/methane ratio, HAWK measurements is a suitable on-line detection technique, due to its simple operation as compared to automated GC. The FTIR in combination with Bayesian statistics is a promising method to determine VOC emission rates, though (due to the relatively high detection limits) the method is more adequate for on plant measurements.

Finally, it is concluded that the objectives of the project: to develop and demonstrate a methodology to assess VOC emissions, and to estimate the contributions by various source categories in a complex industrial area based upon ground-level VOC measurements have been achieved. Presently, there is no need for further development of the method, but there is definitely need for more practical experiences.

#### dissemination

The method was discussed with (inter)national experts at two workshops organised at TNO in Apeldoorn (the Netherlands). The outcome of the project was also presented in the GENEMIS workshop at 7<sup>th</sup> October 1997 in Stuttgart in the framework of the EUROTRAC programme. Publication of the results in relevant magazines and a scientific journal are envisaged in 1998. In addition, TNO will prepare in 1998 an information sheet to be distributed to private companies, industrial associations and relevant authorities to support further application of the method.

## recommendations

It is recommended to apply the method by various potential users such as:

- A local industrial authority may use the method to identify "hot spots" of VOC emissions in an industrial area *without* an Emission Inventory in order to develop a cost-effective abatement strategy in consultation with local industrial companies;
- National or regional enforcement agencies may use the method as a monitoring instrument to validate VOC emission data of the Emission Inventory from a complex industrial area in order to assess the effects of VOC emission reduction measures; and
- Private companies (chemical base, storage and refineries) may use the method (eventually in combination with the FTIR method) to assess diffuse emissions at plant scale in order to control production losses and improve maintenance schemes of for example storage tanks.

10 van 87

TNO-MEP - R 97/435

## Contents

1. INTRODUCTION
1.1 BACKGROUND.       15         1.2 SHORT-TERM OBJECTIVES.       16         1.3 PROJECT ACTORS.       17         1.4 SET-UP OF THE REPORT.       17
2. PROJECT APPROACH
2.1 PROJECT METHODOLOGY       19         2.2 PROJECT ACTIVITIES       21
3. RESULTS OF PHASE I
3.1 SELECTION OF THE PROJECT AREA       25         3.1.1 Spatial distribution of emission sources in Botlek/Pernis       26         3.1.2 C2-C8 VOC emissions by source categories in Botlek/Pernis       26         3.2 DEVELOPMENT OF THE MODELLING STRATEGY       28         3.2.1 Introduction       28         3.2.2 Factor Analysis       29         3.2.3 Spatial Linear Regression Method       29         3.2.4 Underdetermined problems in spatial linear regression: the Bayesian approach       30         3.3 DEVELOPMENT OF THE MEASURING STRATEGY       30
4. RESULTS OF PHASE II
4.1 INTRODUCTION       35         4.2 METEOROLOGICAL CONDITIONS       36
4.3 CORRECTION FOR BACKGROUND AND TRAFFIC EMISSIONS FOR PASSIVE AND ACTIVE SAMPLING 37
4.3.1 Passive sampling results: styrene and benzene
<ul> <li>4.3.2 Direct monitoring (styrene and benzene) and active sampling (ethene) results 39</li> <li>4.4 STYRENE</li></ul>
4.4.2 Quality check of active sampling data by comparing with EI data 40
4.4.3 Estimate of styrene emission rates using spatial linear regression and passive measurements
4.4.4 Estimate of styrene emission rates using spatial regression and_active measure-
4 4 5 Discussion and conclusions 44
4.5 BENZENE
4.5.1 Quality check of passive sampling data by comparison with EI data
4.5.2 Quality check of active sampling data by comparing with EI data
4.5.3 Estimate of benzene emission rates using Bayesian statistics and passive measure-
ments 47

12 van 87

TNO-MEP - R 97/435

4.5.4 Estimate of benzene emission rates using Bayesian statistics and activ	ve measure-
ments	51
4.5.5 Comparison of benzene emission rates computed from passive and a	ctive sampling53
4.5.6 Discussion and conclusions	
4.6 C <sub>2</sub> -C <sub>5</sub>	57
4.6.1 Factor analysis	57
4.6.2 C <sub>2</sub> -C <sub>5</sub> emission rates by FTIR measurements	62
4.6.3 $C_2$ - $C_5$ air concentrations by HAWK measurements	69
5. CONCLUSIONS AND RECOMMENDATIONS	71
6. DISSEMINATION ACTIVITIES	
7. REFERENCES	77
8. AUTHENTICATION	87

Annex 1 Results of the measurements

# List of abbreviations

EC	European Commission
EI	Emission Inventory
FTIR	Fourier Transform Infra Red
DCMR	Environmental Protection Agency of Rijnmond
KNMI	Royal Netherlands Meteorological Institute
KWS-2000	Netherlands Programme to reduce VOC emissions
TNO	Netherlands Organisation for Applied Scientific Research
VOC	Volatile Organic Compounds
VROM	Netherlands Ministry for Housing, Physical Planning and Environmental Protection

TNO-MEP - R 97/435

## 1. Introduction

## 1.1 Background

Air concentrations of Volatile Organic Compounds (VOC) can potentially lead to increased concentrations of ground level ozone, which may cause respiratory problems, damage to vegetation and contribute to the greenhouse effect. Next to traffic, various industrial activities are regarded as important sources for air pollution by VOC. Consequently, in the vicinity of large-scale industrial activities VOC concentrations may directly affect human health.

At European level, measures are introduced to reduce VOC emissions both from industrial and traffic sources. At national level, programmes are also initiated to reduce VOC levels in ambient air. For example, in the Netherlands the "KWS-2000" project aims at a reduction of VOC emissions in the year 2000 with 50% compared to the year 1981.

In order to evaluate the effect of emission reduction measures within industrial areas, one requires an answer to the question: Which VOC sources in an industrial area are contributing how much to ambient air concentrations in the vicinity of the industrial area? This problem is illustrated in Figure 1.1.



Figure 1.1 Air pollution around an industrial area.

Figure 1.1 shows VOC emissions by various sources within an industrial area from low stacks (1), diffuse emissions from storage tanks (2) and high stacks (3). After plume rise and dispersion in the atmosphere, VOC emissions result in increased VOC ground-level concentrations. In the Netherlands, the National Emission Inventory (EI) is used to collect information on VOC emissions and subsequently, in combination with dispersion models, ground-level air concentrations of VOC are computed.

TNO-MEP - R 97/435

16 van 87

The EI includes registration, location and presentation of emission data of both industrial and non-industrial sources in the Netherlands. The emission data are updated every year by TNO under a contract by National Authorities. Emissions to air and water are distinguished into *industrial source* and *collective source categories*. The central database of EI contains emissions of about 900 compounds from 700-800 major companies. For large industrial point sources the emissions are based upon detailed information of each individual plant. While emissions of small and medium-sized enterprises, as well as non-industrial sources are based upon economic data and emission factors.

However, there are constraints to compute the effect of air pollution reduction measures based upon EI data:

- firstly, several studies have indicated discrepancies between the "EI-based" methodology and actual ambient air measurements, which may be due to incorrect emission factors and/or un-know sources such as diffuse emissions;
- secondly; ambient air concentrations estimated from EI data concern annual average concentrations and hence, there is at least a delay of one year to provide these concentrations and no information is provided on fluctuations within a year for example during a smog-episode; and
- thirdly; EI data do not exist in all European countries.

For the aforementioned reasons, complementary to the "EI-based" methodology, there is a need for an alternative method to assess VOC emissions from large-scale industrial areas. Such an instrument may be applied i.) in European countries without EI data to monitor the effects of VOC emission reduction measures, and ii.) in countries with EI data to validate EI data, to identify unknown VOC sources and to collect more detailed and readily available information on (for example) monthly VOC emission rates.

In the framework of this project, a methodology has been developed based upon VOC ground-level air measurements to identify and quantify the contribution by various VOC emission sources from an industrial area.

## 1.2 Short-term objectives

The short term objectives of the project are:

- to develop a methodology to assess VOC emissions based upon VOC groundlevel measurements;
- to demonstrate the methodology in the Rijnmond area by identification of various sources of VOC emissions and computation of the strength of their emission rates

## 1.3 Project Actors

TNO has been granted by the European Commission under the LIFE programme to implement the project. The project has been monitored by Mr. M. Leijendeckers of Tauw Milieu on an assignment by the European Commission.

The Netherlands Ministry of Housing, Physical Planning and the Environment (VROM), the Environmental Protection Agency of Rijnmond (DCMR) and SHELL Research Ltd., the Netherlands have supported the project by the following contributions:

- VROM contributed NLG. 250.000,-. In the framework of the VOC-emission reduction programme "KWS-2000" in the Netherlands, VROM is especially interested in an instrument to assess VOC emissions by large-scale industrial sources. The generated data may be compared with VOC emission data provided by the Netherlands Emission Inventory in order to identify "unknown" and diffuse sources.
- Under the project, DCMR has made available three of its monitoring sites to install sampling and on-line equipment, as well as air quality data collected under its routine monitoring programme. The contribution by DCMR has been estimated to be equivalent with NLG. 100.000 based upon the reduction of the originally planned inputs by TNO (i.e. staff, analysis, equipment and travelling) in phase II.
- SHELL is one of the major industries in the Rijnmond area with large-scale chemical operations, storage of oil products and refineries. SHELL Research Ltd., the Netherlands contracted SHELL, United Kingdom to perform remote sensing measurements and will make available the collected data to the project. The contribution by SHELL has been estimated to be equivalent with NLG. 100.000 based upon the reduced, planned inputs by TNO (i.e. staff, analysis, equipment and travelling) in phase II.
- TNO has managed implementation of the project and contributed by expertise, staff, equipment, chemical analytical services and models.

## 1.4 Set-up of the Report

In Chapter 2 the project approach and planning of activities are described. Chapter 3 provides information on the results of phase I related to development of the methodology and selection of the pilot area within Rijnmond. Chapter 4 presents the results of phase II are presented related to demonstrate the methodology in the Botlek/Pernis area. In Chapter 5, conclusions and recommendations are discussed. Chapter 6 describes dissemination activities, Chapter 7 presents authentication of the report and Chapter 8 includes references. Annex 1 gives details on the results of the measurements, while staffing and financial aspects of the project are reported in a separate document.

18 van 87

TNO-MEP - R 97/435

## 2. Project approach

## 2.1 Project methodology

In order to elaborate the project methodology, some background is provided on the general methodology to determine emission rates from ground-level measurements. This is illustrated by the single-source and multiple-sources situation, respectively

#### Single source situation

A Gaussian-plume model can be applied for "inverse dispersion" modelling to calculate the VOC emission rate from ground-level VOC air concentration measurements. This is relatively simple in case of a single stack and two monitoring sites up- and downwind of the emission source. The difference between VOC air concentrations measured at the two monitoring stations during an appropriate wind direction is attributed to the VOC emission source. Using the Gaussian-plume model and adequate meteorological conditions to account for the plume rise and atmospheric dispersion, the emission rate can be calculated.

#### Multiple-sources situation

The situation is more complex in an industrial area such as the Botlek/Pernis area, which combines various point (i.e. high and low stacks) and diffuse sources (i.e. evaporation from storage tanks, spillage and traffic). The contribution of the multitude of these VOC sources can be measured by two alternative measurement approaches by so-called *receptor* and *emission* locations. These two approaches are applied in this project and are illustrated in Figure 2.1.



Figure 2.1 Measurement strategy in and around an industrial area.

TNO-MEP - R 97/435

20 van 87

Figure 2.1 shows VOC emissions within an industrial area by low stacks (1), diffuse emissions from storage tanks (2) and high stacks (3). VOC emissions result in increased VOC ground-level concentrations down-wind at location (R2) as compared to the up-wind location (R1). These monitoring locations *outside* the industrial area concern *receptor* monitoring sites. Also *within* the industrial area ground-level VOC air concentrations are increased at for example locations E1, E2 and E3. Monitoring stations *inside* the industrial area relate to *emission* monitoring sites. The two measurement approaches used in this project concern receptor and emission sites:

- a triangle of three *receptor* locations around the area to maximise the collection of information on dispersion of the multitude of sources. The measurements provide hourly ground-level concentration differences at six wind directions in which one station serves as receptor station, while the opposite station is the background station. The measured concentration difference at appropriate wind directions now consists of the sum of contributions of several VOC emission sources; and
- fourteen *emission* locations have been selected to measure two-weekly ground-level air concentrations with passive samplers. In this case, the air concentration differences between the fourteen locations and the six sampling periods, as well as the meteorological conditions provide information on VOC emissions.

The next step in both cases is to apply factor analysis on the measured air concentrations (passive sampling) or differences of measured air concentrations (active sampling) to identify the contributing sources. The next step is "spatial linear regression" or "Bayesian statistics" in combination with inverse dispersion modelling is used to calculate the emission rates of the identified emission sources. The determined emission rates are compared to EI data to validate the approach. Finally, the results from both the two measurement approaches are compared, in order to recommend the most cost-effective methodology to estimate VOC emission sources by ground-based measurements.

#### remote sensing measurements

In addition to the monitoring stations at the three receptor and fourteen emissions sites, also remote sensing measurements have been applied at two receptor sites and at several emissions sites within the industrial area. Remote sensing concerns direct optical air concentration measurements over an "open path" length and provides information on plume profiles (ratio between air concentrations of specific hydrocarbons) of VOC emissions by typical industrial activities. Using Bayesian statistics results in quantifying emission rates of VOC emission sources. Remote sensing measurements are applied most extensively on industrial plants to determine VOC emissions for example during different production conditions. In this project, the application of remote sensing to estimate VOC emissions from a large-scale industrial area is tested.

The aforementioned steps in the project methodology are further elaborated below:

- the first step is to select within the Rijnmond region, which has a size of about 40\*20 km, a smaller pilot area to demonstrate the methodology. The selection is based upon considerations such as the size of the area, diversity of VOC sources and existing monitoring sites of the local environmental protection agency;
- the second step is to develop a modelling strategy in order to calculate emission rates from measured ground-level monitoring sites. This requires an one-hour, inverse dispersing model in combination with regression analysis to calculate emission rates from ground-based measurements. The outcome of this step is a modelling strategy and recommendations on the measuring strategy (frequency of sampling; number of components; number of monitoring locations; required meteorological data etc.);
- the third step is to collect a-prior information on the selected industrial area (type of industrial activities, the height and location of emission sources area and estimated VOC emission rates) in order to select the locations of *receptor and emission* sites. Subsequently, sampling and measuring equipment are installed and appropriate meteorological data collected;
- the fourth step is to demonstrate the methodology by implementing measurements followed by data analysis using the developed modelling strategy. This step results in multiplication factors for "default emission rates" of VOC emission sources from the EI and emission rates of unknown VOC sources identified in this step;
- the fifth step is to compare the results of both emission and receptor sites, and remote sensing measurements with EI data, to discuss the methodology with (external) experts and to disseminate the demonstrated methodology.

These aforementioned five basic steps in the project methodology have been used to guide project activities, which are further elaborated in the next section.

## 2.2 Project activities

The project duration was two years from January 1996 till February 1998. The project has been implemented following a phased approach:

*Phase I* was the Research and Development phase and was implemented from 1/1/1996 to 1/4/1997. Phase I was directed to develop a modelling methodology, to select a monitoring strategy and to consult national and international experts. During this phase, literature has been surveyed and models have been tested. The outcome of this phase was discussed with experts during the workshop in March 1997. This phase has been finalised by submitting the Interim Report.

22 van 87

TNO-MEP - R 97/435

In Table 2.1, the activities in phase are presented in an activity bar chart.

Table 2.1 Activity bar chart of activities implemented in phase I..

STEP	ACTIVITY	100	1997							
		Feb	Apr	Jur	1	Aug	Okt	Dec	Feb	
pilot area	- consultation with local authority									
selection	- El data collection									
	- dispersion calculations									
	- profile calculations		104 30							
	- pilot area defined									
modelling	- define specifications			13						
strategy	- test various models						1000			
	- apply models on El data of pilot area									
	- define specifications for measurements									
monitoring	- consultation DCMR									
strategy	- inspection of potential monitoring sites									
	- establish QA programme								168	
	- select sites, methods and frequency									
dissemination	- workshop									·
reporting	- Draft Interim Report									•
	- Final Interim report									•

The results of phase I are described in detail in Chapter 3.

*Phase II* was the Demonstration phase and has been executed from 1/4/1997 to 1/2/1998. Phase II was related to actual measuring activities in the Bot-lek/Pernis area, to process the collected data in accordance to the modelling strategy and to report the outcome of the project. This phase has been concluded with a workshop in November 1997 to disseminate the results of the project and drafting of the Final Report.

In Table 2.2, the activities in the project are presented in an activity barchart.



The results of phase II are described in detail in Chapter 4.

Table 2.2 Activity bar chart of activities implemented in phase II.

24 van 87

TNO-MEP - R 97/435

## 3. Results of Phase I

## 3.1 Selection of the project area

The project area was envisaged within the Rijnmond region. Rijnmond is located along the river Rhine in the west of the Netherlands in between Rotterdam and the North-Sea. It contains among other the harbour of Rotterdam, storage of oil products and industrial activities by refineries and chemical base plants. The size of the area is approximately 40 \* 20 km, which is too large and too complex to develop a monitoring methodology. The selection of the project area was based upon location of VOC sources, variety in high and low stacks, presence of storage tanks, size of the area, strength of VOC rates, existing monitoring sites of the local enforcement authority "DCMR" and the modelling methodology. Based upon these criteria and model calculations of various potential areas, it was decided to select the centre area of Rijnmond, the so-called Botlek/Pernis area with a surface of 4\*2 km. The Botlek/Pernis area, which is enclosed by DCMR's and project monitoring stations at Maassluis, Zwartewaal and Hoogyliet is presented on a map in Figure 3.1



Figure 3.1 The Botlek/Pernis area and three receptor monitoring sites.

The Botlek/Pernis area is defined by the co-ordinates: 433 - 435 (South-North) and 78 - 82 (West-East).

TNO-MEP - R 97/435

26 van 87

In the following sections, information is presented on the spatial distribution of VOC emissions and most important sources in Botlek/Pernis. This information has been retrieved from the Netherlands EI and has been used to design the measurement strategy, as well as to support the interpretation of the modelling exercise. It is noted that in case of industrial areas without an EI database, the aforementioned information is collected from site-inspection and "default" emission rates of VOC compounds may be estimated from similar industrial ac-tivities elsewhere.

#### 3.1.1 Spatial distribution of emission sources in Botlek/Pernis

EI data are used to illustrate VOC emissions in Botlek/Pernis by various source categories. In Figure 3.2, the spatial distribution of four important "individual industrial" source categories are presented.



Figure 3.2 Spatial distribution of four "individual industrial" source categories in Botlek/Pernis.

## 3.1.2 C<sub>2</sub>-C<sub>8</sub> VOC emissions by source categories in Botlek/Pernis

The EI contains more than 900 compounds emitted by various sources in Botlek/Pernis. It is not possible to involve more than 900 compounds in this project and therefore, VOC compounds have been limited to " $C_2$ - $C_8$ " hydrocarbons. In section 3.3 more details are provided on the selection of the compounds.

The most important sources of the selected " $C_2$ - $C_8$ " VOC emissions by various source categories in Botlek/Pernis are presented in Table 3.1.

Compound	Hydro- carbon code	Emission	Contribution of source categories to VOC emissions (%)							
		(ton/y)	S1	S2	<b>S</b> 3	<b>S</b> 4	11	12	13	others
Ethane	C <sub>2</sub>	239	48	0	1	3	31	3	12	3
Ethene	C <sub>2</sub>	1020	62	0	3	5	18	9	2	1
Acetylene	C <sub>2</sub>	890	73	0	0	4	0	22	0	0
Propane	C <sub>3</sub>	749	52	0	0	3	29	5	9	2
Propene	C <sub>3</sub>	613	50	0	1	з	4	40	2	0
n-Butane	C4	2003	6	0	2	0	57	4	27	4
i-Butane	C <sub>4</sub>	672	9	0	1	0	47	10	27	4
n-Pentane	C <sub>5</sub>	1503	10	0	2	1	51	3	29	4
i-Pentane	C <sub>5</sub>	1228	25	0	2	1	46	2	22	2
Hexane (tot.)	C <sub>6</sub>	2197	25	0	7	1	37	з	25	2
Benzene	C <sub>6</sub>	629	58	0	1	6	13	10	12	0
Heptane (tot.)	C7	998	30	0	1	2	37	2	26	2
Toluene	C7	1697	47	0	11	3	18	10	10	0
Styrene	C <sub>8</sub>	446	14	0	2	1	0	80	3	0
Xylene (tot.)	C8	794	73	1	6	4	0	16	0	0
Octane (tot.)	C <sub>8</sub>	1033	52	0	1	3	23	2	17	2

Table 3.1 Contribution of source categories to the total emission of " $C_2$ - $C_8$ " hydrocarbons in Botlek/Pernis

In Table 3.1, the first four emission categories concern "collective sources": S1="traffic"; S2="navigation"; S3= "small companies" and S4="other emissions per inhabitant". While, the last four emission categories refer to "individual industrial sources" : I1="petrochemical industries"; I2="Chemical base industries" and I3="storage companies" and "other"= the sum of VOC emissions by other industrial sources.

From Table 3.1, it is noted that traffic (S1) is an important emission source for VOC. In the data analysis industrial emissions have been corrected for VOC emissions by traffic. In addition, it is noted that more than 80% of the industrial VOC emissions in the Botlek/Pernis area are produced by three industrial source categories: refineries (I1), chemical base (I2) and storage tanks (I3).

Based upon the aforementioned information, it was concluded in Phase I of the project that the Botlek/Pernis area is representative for a large number of industrial areas in Europe and therefore, an appropriate area to demonstrate the methodology.

TNO-MEP - R 97/435

28 van 87

## 3.2 Development of the modelling strategy

### 3.2.1 Introduction

The objective of the modelling strategy is to calculate from ground-level air concentration measurements, the VOC emissions in the Botlek/Pernis area. This includes a number of steps: identification of VOC source categories, identification of locations and computation of emission rates of the identified sources. Various models and statistical tools are available to address these steps. In Phase I of the project, using emission rates of the EI and meteorological data from former years, various modelling strategies were applied on the Botlek/Pernis area. This resulted firstly in a modelling strategy to analyse the data collected by actual measurements and secondly provided conditions for the measurement strategy.

The basic elements in the developed modelling strategy are a combination of *dispersion* and *receptor* models:

- Forward and inverse dispersion modelling; In forward-dispersion modelling, ground-level air concentrations are calculated using emission rates and meteorological data. In inverse-dispersion modelling, emission rates are computed from measured air concentrations at ground-level and meteorological data. Hence, inverse-dispersion modelling is the opposite of forward-dispersion modelling and vice versa. A Gaussian-plume model describes the relation between emission rates, air concentrations and meteorological conditions. In this project, the 1997 up-dated version of the Netherlands National Plume model has been used both for hourly forward and inverse modelling. In the developed modelling strategy, forward-dispersion modelling is the first exercise to compare the measured with calculated air concentrations using EI emission data, which provides a first order estimate on the reliability of the measurements. Inverse-dispersion modelling in combination with receptor modelling is the final step in the modelling strategy to compute emission rates;
- Receptor models generate information on emission sources by measured air concentrations at receptor locations. In this project, factor analysis, spatial linear regression and Bayesian statistics have been used. Receptor models are mainly applied to identify sources ("factor analysis") and to estimate their contribution to air concentrations at receptor locations ("spatial linear regression" and "Bayesian statistics"). In our methodology, the latter step has been combined with inverse-dispersion modelling in order to compute emission rates of the identified VOC sources. In view of the importance of factor analysis, spatial linear regression and Bayesian statistics in the modelling approach, these methods are briefly reviewed in the next three sections.

## 3.2.2 Factor Analysis

The aim of factor analysis is to identify individual sources or source categories, which contribute to air concentrations at receptor locations. Prior information on potential sources (i.e. emission profiles and locations) further supports this identification.

In this project, factor analysis has been used as an instrument for an exploratory analysis of the data to identify various VOC sources. Software for factor analysis is available under standard statistical packages.

#### 3.2.3 Spatial Linear Regression Method

Spatial linear regression results in information on the contribution of various sources to the air concentration of a single compound at various receptor locations. Hence, it provides similar results as the well-known Chemical Mass Balance (CMB) method, but main differences are that spatial linear regression uses air measurements of a single compound at various receptor locations (while CMB uses air concentrations of more compounds at one receptor location). The advantage of spatial linear regression over CMB is that there is no need for information on emission profiles.

The spatial linear regression method is based upon linear regression as applied by Mulholland and Seinfield (1995). The regression procedure requires a number of *measured* concentrations at receptor locations as well as *calculated* concentrations at the same locations by using forward-dispersion modelling, "default" EI emission rates and actual meteorological conditions. By linear regression the "default" EI emission rates are varied to give the closest resemblance between the *measured* and *calculated* concentrations. The method results in *modified* emission rates.

An important condition for the method is that the number of monitoring locations exceeds the number of emission sources. In that case the regression is over-determined and can give reliable results. For example for styrene (as will be illustrated in Chapter 4) these conditions are fulfilled, as styrene emissions are dominated by only two sources.

However, for most components in a complex industrial area such as the Botlek/Pernis area, there are a larger number of emission sources. Consequently, it is not possible to apply the relatively simple spatial linear regression method to estimate source strength of a multitude of sources and therefore, a more advanced technique is required based upon the Bayesian statistics. 30 van 87

TNO-MEP - R 97/435

# 3.2.4 Underdetermined problems in spatial linear regression: the Bayesian approach

As mentioned above spatial linear regression requires more known (measured concentrations) than unknown (emission rates) variables. However, in the opposite case it is still possible to extract information on emission rates from measured concentrations by using prior information present in the "default" EI emission rates. This is done by using Bayesian statistics [Buck B., Macaulay V.A. (1991): "Maximum entropy in action"].

The method calculates *modified* emission rates (starting with "default" emissions rates) from the measured concentrations. The method also calculates *uncertainties* on the modified emission rates. If the measurements strongly are determined by the emission of a certain source the uncertainty will be low. While emissions from sources with low rates and/or at large distances are only weakly represented by the measurements and therefore, the uncertainty on the most probable emission rate is large.

The aforementioned forward- and inverse-modelling and statistical tools (factor analysis, spatial linear regression and Bayesian statistics) have been tested in Phase I of the project. In Chapter 4, application of the modelling strategy is demonstrated on the measured air concentrations in and around the Bot-lek/Pernis area. The next activity in Phase I of the project was to develop an appropriate measurement strategy.

## 3.3 Development of the measuring strategy

Based upon the required inputs for the modelling strategy, the following measuring strategy was developed, which concerns the following aspects:

compounds; The demonstrated methodology is valid only when the relation between emission and ground-level concentration of the compounds is linear, i.e. when the effects on the concentration of deposition and chemical conversion during transport from the source to the monitoring location are small. A second consideration in selecting hydrocarbons is the interest of policy makers and/or enforcement agencies in certain hydrocarbons, including: i.) important ozone precursors the "C<sub>2</sub>-C<sub>5</sub>" hydrocarbons, ii.) a toxic compound such as benzene (the "C<sub>6</sub>" hydrocarbon), which is emitted both by traffic and evaporation from storage tanks and iii.) styrene (the "C<sub>8</sub>" hydrocarbon) which is emitted mainly from high stacks by industrial activities. These considerations have resulted in the selection of the following "C<sub>2</sub>-C<sub>8</sub>" hydrocarbons: ethane, ethene, acetylene, propane, n-butane, i-butane, n-pentane, i-pentane, total hexane, benzene. These compounds were measured on-line with gaschromatography or off-line analysed in TNO's

certified analytical laboratory. Shell UK implemented the remote sensing measurements.

- monitoring period; The monitoring period was from mid-April till mid-July 1997. A longer period was not possible due to time restrictions in the project period. A more reliable comparison with annual EI emission rates requires measurements for at least a year. However, the three months measuring campaign in this project represents a sufficient long period to demonstrate the methodology.
- receptor sites; ground-level air concentrations are measured at three locations at a triangle up- and downwind of the source area (the so-called "receptor sites") in order to determine the total VOC-emissions from the industrial area. The selected receptor sites are Zwarte Waal, Maassluis and Hoogvliet, which were already indicated in Figure 3.1. The distance to Botlek/Pernis is in the order of 5-10 km. The sites have been operated in cooperation with DCMR, as these sites are part of the routine monitoring network of DCMR. At the receptor sites, "C<sub>2</sub>-C<sub>8</sub>" hourly measurements have been performed by on-line monitoring and active sampling equipment. Hourly measurements are the shortest period for most of the selected hydrocarbons to be monitored in ambient air. These measurements are relatively costly, require weather-proof housing and electricity supply, and are prone to vandalism. These measurements concern the following details:
  - Three continuous so-called "BTX-monitors" have been operated at the three receptor locations. Benzene, toluene, m-xylene, o-xylene, pxylene, ethyl benzene and styrene ("C<sub>6</sub>-C<sub>8</sub>") have been measured at 30minutes sampling averages;
  - ♦ A so-called "C<sub>2</sub>-C<sub>5</sub> monitor" (an automated GC) has been installed at Zwarte Waal. At the other two sites hourly samples in "Tedlar" bags have been taken for one hour at alternating hours every day between 10.00 AM and 15.00 h PM. This provides information on the diurnal variation of "C<sub>2</sub>-C<sub>5</sub>" air concentration differences between the three sites;
  - Each day at all the three sites hourly "C<sub>6</sub>-C<sub>8</sub>" samples have been collected by "Tenax" absorption tubes at the same time as the aforementioned "C<sub>2</sub>-C<sub>5</sub>" samples;
  - At the three sites, monthly samples have been collected by passive samplers to measure "C<sub>6</sub>-C<sub>8</sub>" air concentrations; and
  - Remote sensing measurements were performed by a "Hawk monitor" (Siemens) with an open path length of 500 m for two weeks in Hoogvliet and Maassluis to measure the sum of "C<sub>2</sub>-C<sub>5</sub>" air concentrations. It is a optical method with a light source and reflection of the light beam on a reflector. The amount of light absorbed at specific wavelength is a measure of the air concentration of a specific compound. The goal of these continuous measurements was to compare the HAWK performance with the aforementioned GC measurements to detect diurnal fluctuations of "C<sub>2</sub>-C<sub>5</sub> VOC" emissions. The advantage of the HAWK monitor is that

32 van 87

its simple to operate under ambient air conditions compared to on-line GC measurements.

• *emission sites;* ground level concentrations were also measured within the source area in order to determine VOC-emissions with emphasis on *diffuse* and low-stack sources. The emission sites were fourteen locations within Botlek/Pernis near the most important " $C_6$ - $C_8$ " emission sources as identified by the Netherlands Emission Inventory. The fourteen locations are presented in Figure 3.3.



Figure 3.3 The fourteen locations of the passive samplers within the Botlek/Pernis area.

At the emission sites air concentrations of "C6-C8" have been collected by passive samplers by two-weekly measurements. Passive samplers are cost-effective and can be easily installed at many locations (without electricity supply) and are rarely removed during the sampling period. Two weeks are the shortest period for most components to be monitored by passive sampling in ambient air. Hence, it is difficult to link the measurements with wind directions, but it is demonstrated in this project that this disadvantage is compensated by the large spatial coverage of passive samplers. Unfortunately, no passive samplers are available for "C2-C5" components. Details of the passive samplers are as follows:

Active carbon-coated passive samplers are extracted after sampling with carbondisulfide and analysed with GC. The sampling periods of the passive samplers over a total period of three months are presented in Table 3.2.

period	start	end			
1	10-04-97 16:00	1-05-97 15:00			
2	1-05-97 15:00	15-05-97 10:00			
3	15-05-97 10:00	29-05-97 15:00			
4	29-05-97 15:00	12-06-97 14:00			
5	12-06-97 14:00	26-06-97 12:00			
6	26-06-97 12:00	10-07-97 12:00			

Table 3.2 The six sampling periods of the fourteen passive samplers.

In total 90 air measurements were performed with the passive samplers. In addition to the passive samplers, remote sensing was conducted during one week to measure directly " $C_2$ - $C_5$  VOC" air concentrations with a Fourier Transfer Infra Red (FTIR) instrument. FTIR is based upon a infra red beam which is reflected in ambient air on a "mirror". VOC components present in the beam absorb infrared radiation and the relative contribution of each component (and thus the air concentrations) can be specified. In combination with meteorological data, a dispersion model and Bayesian statistics, a map is produced with emitting " $C_2$ - $C_5$  VOC" sources up-wind of the path length. In addition to the location of emission sources and strength of emission rates, also the height of the emission sources is calculated. During the one-week campaign in July 1997, 21 beam positions with a path length of 200 - 300 m were used within the Botlek/Pernis area. The beam positions are indicated in Figure 3.4.



Figure 3.4 The beam positions of the FTIR measurements in the Botlek/Pernis area.

34 van 87

TNO-MEP - R 97/435

• *meteorological data;* Meteorological data have been retrieved from the National Meteorological Network at two locations: Airport Rotterdam (about 10 km north-east from the Botlek/Pernis area) and Geulhaven at the centre of Botlek/Pernis. The latter only provides windspeed and -direction, while at the former a "complete" set of meteorological parameters are collected, as required for the short-term National Plume model.

In the next chapter the results of the measurements and modelling are presented in order to demonstrate the methodology.

## 4. Results of Phase II

#### 4.1 Introduction

The results of the VOC air concentration measurements at the three receptor sites: Hoogvliet, Maassluis and Zwartewaal, and the fourteen emission sites are enclosed in Annex I. Due to reporting restrictions, *the presentation of the modelling results is limited to styrene, benzene and* " $C_2$ - $C_5$ " hydrocarbons with emphasis on ethene. These components best represent the methodology for the following reasons:

- styrene, as the number of emission sources is limited and therefore this compounds illustrates a relatively simple situation to compute emission rates;
- benzene has been selected because of its health effects and the large number of low and high emission sources; and
- "C<sub>2</sub>-C<sub>5</sub>" emissions are related to the ozone-forming potential of these compounds. "C<sub>2</sub>-C<sub>5</sub>" hydrocarbons include ethane, ethene, acetylene, i,n-butane and i,n-pentane. For presentation reasons, the case for "C<sub>2</sub>-C<sub>5</sub>" is focused on ethene.

In the following five sections of this chapter, *firstly* the meteorological conditions during the measuring period are presented (section 4.2), the *second section* concerns correction of the measured air concentrations for large-scale transport of "background" polution and traffic emissions (section 4.3), and followed by *three sections* on *styrene (section 4.4), benzene (section 4.5)* and  $C_2$ - $C_5$ , with emphasis on ethene (section 4.6). The presentation in the latter three sections 4.4-4.6 follows a schematic approach:

- firstly, ground-level air concentrations at the receptor and emissions sites are estimated by forward-dispersion modelling using actual meteorological conditions and emission rates of the Emission Inventory of 1995. The results are compared with actual measurements of ground-level air concentrations of styrene, benzene and ethene. The outcome provides a "first order" estimation of the reliability of the measurements;
- secondly, based upon the ground-level measurements of both the receptor and emission sites, the meteorological conditions during the measurements and the modelling strategy, styrene, benzene and ethene emission sources are identified and subsequently the emission rates for each compound is calculated. This is the actual demonstration of the methodology.; and
- thirdly, each section is finalised with conclusions.

36 van 87

## 4.2 Meteorological conditions

Average meteorological data from half April till half July 1997 at Airport Rotterdam concern: temperature 14 °C, wind speed 4 m/s, wind direction south/south-east and rain intensity 0.1 mm/h. These data reflect "normal" meteorological conditions for this season in the Netherlands. Wind directions during the sampling period is a critical meteorological parameter for transport of air pollutants from emission sources to monitoring sites. In Figure 4.1 frequency distributions are presented of wind directions at each of the six two-weekly sampling periods of passive samplers in the Botlek/Pernis



Figure 4.1 Frequency distribution of wind directions at Botlek/Pernis in the period half April to half July 1997 distinguished in six two-weekly periods.
Figure 4.1 shows that during the six two-weekly sampling periods for the five periods: 1-4 and 6, more than 50% of the time the wind direction is concentrated in a window of about 20 degrees. Consequently, air concentrations measured by passive sampling (with a sampling period of two weeks) are better correlated with VOC emissions in the up-wind direction than a-prior expected.

## 4.3 Correction for background and traffic emissions for passive and active sampling

### 4.3.1 Passive sampling results: styrene and benzene

In order to assess the contribution of *industrial* VOC emissions from the Botlek/Pernis to air concentrations measured by passive samplers at the 14 sites within the Botlek/Pernis area, the measured concentrations need to be corrected for contributions by the "background" and traffic emissions. *Background concentrations*; The large-scale "background" has been determined by hourly measurements using "Tenax" sampling tubes at the three receptor sites. Each site probes a wind sector with wind directions directed away from the Botlek/Pernis area. Averages for these sectors have been calculated for styrene and benzene. Using the distribution of the windsectors for all six periods of the passive samplers (see Figure 4.1), background concentrations have been calculated by weighted avarages. Table 4.1 shows the results for benzene and styrene.

Table 4.1	Background air concentrations of benzene and styrene in $\mu g/m^3$ at the recep-
	tor sites Maassluis, Hoogvliet and Zwartewaal at three background wind-
	sectors (315-90, 90-215 and 215-315) for the six periods of passive sam-
	pling.

	Maassluis 315 - 90	Hoogvliet 90 - 215	Zwartewaal 215 - 315	benzene (μg/m³)	styrene (μg/m³)
period 1	59%	13%	28%	1.3	0.4
period 2	19%	45%	36%	1.6	0.3
period 3	61%	23%	16%	1.4	0.4
period 4	81%	16%	3%	1.4	0.4
period 5	31%	30%	39%	1.4	0.3
period 6	46%	39%	16%	1.6	0.4

The results show that variation of the *background* concentrations of benzene and styrene at various wind directions for the six periods is small.

TNO-MEP - R 97/435

38 van 87

*Traffic contribution*; The measured air concentrations have also been corrected for the contribution of traffic emissions. The profile of traffic emissions has been determined by TNO in Berlin and Rotterdam in earlier studies in 1996/97. The profile of traffic emissions includes all components measured by the passive samplers in Botlek/Pernis. Comparison of the traffic profiles and the profiles of the air concentrations measured by the passive samplers indicates that the contribution of 123-trimethylbenzene by industrial sources in the Botlek/Pernis area is negligible. *Therefore, it is assumed that the air concentration of 123-trimethylbenzene in the Botlek/Pernis area is determined for 100% by traffic emissions*. Using the traffic profile and specifically the ratio benzene/123-trimethylbenzene and the ratio styrene/123-trimethylbenzene, the traffic contributions to benzene and styrene have been calculated for each of the six periods and for each of the fourteen pasive samplers. The results for benzene are shown in Table 4.2.

Table 4.2	Traffic contribution to benzene air concentrations at the 14 locations (P1-
	14) in the Botlek/Pernis area during six periods (period 1-6) in $\mu g/m^3$ .

	period 1	period 2	period 3	period 4	period 5	period 6
P1	0.00	0.00	0.00	0.01	0.00	0.00
P2	0.00	1.79	0.00	1.63	0.00	0.00
P3	0.00	0.00	0.00	0.19	0.00	0.00
P4	0.38	1.13	0.39	1.33	0.13	0.13
P5	0.56	0.00	0.26	0.86	0.00	0.00
P6	2.23	1.84	2.94	2.28	1.82	1.13
P7	1.41	0.78	1.15	1.19	0.74	0.15
P8	0.64	0.00	0.88	1.98	0.64	0.24
P9	0.52	0.00	0.75	1.19	0.00	0.00
P10	0.44	0.70	1.00	2.11	0.24	0.24
P11	1.66	0.93	2.17	3.56	1.29	1.24
P12	0.99	0.34	1.66	2.93	0.45	0.70
P13	0.97	1.59	1.41	2.67	0.99	0.73
P14	0.44	0.85	0.97	0.79	0.46	0.00

Using the data in Table 4.2, it is calculated that the relative contribution of traffic to benzene concentrations within the Botlek/Pernis area is less than 20% for all concentrations exceeding 5  $\mu$ g/m<sup>3</sup> benzene.

Analogous to the benzene concentrations, also for styrene the traffic contribution has been calculated. For styrene, the results indicated that within the Botlek/Pernis area the traffic contribution to styrene is less than 10 % for all concentrations exceeding 2  $\mu$ g/m<sup>3</sup> styrene. From these results, it is concluded that the air concentrations of benzene and styrene in the Botlek/Pernis area are determined for more than 80 % by local industrial emissions.

In the analysis in the sections 4.4 and 4.5, the "background- and trafficcorrected" air concentrations of passive sampling results are used and hence, only the contributions by the Botlek/Pernis area to the air concentrations at the fourteen sampling locations are taken into account.

## 4.3.2 Direct monitoring (styrene and benzene) and active sampling (ethene) results

The approach to correct for background concentrations and traffic in case of hourly measurements at Zwartewaal, Hoogvliet and Maassluis differs from the described approach on passive sampling. The air concentration measurements at these monitoring locations can be correlated with the wind direction during sampling and hence, relevant wind sectors can be assigned with the Bot-lek/Pernis area in between the up- and down-wind locations. Subsequently, the background and traffic contributions to styrene, benzene and ethene air concentrations have been reduced by using the *differences of the measured air concentrations*. The relevant wind sectors for the three locations are as follows (see also Figure 3.1):

- Hoogvliet; the relevant wind sector is 270 90 with Hoogvliet down-wind of Botlek/Pernis, while the measurements at Maassluis serve as the up-wind location;
- Maassluis; the relevant wind sector is 90 180 with Maassluis down-wind of Botlek/Pernis, while the measurements at Hoogvliet serve as the up-wind location; and
- Zwartewaal; the relevant wind sector is 45-90 with Zwartewaal down-wind of Botlek/Pernis, while the measurements at Maassluis serve as the up-wind location;

In the analysis in the sections 4.4-4.6 of the active sampling results, *the air* concentrations differences in the relevant wind sectors at the three locations Hoogvliet, Zwartewaal and Hoogvliet have been applied. Hence, only the contributions by the Botlek/Pernis area to the air concentrations at the three locations are taken into account.

## 4.4 Styrene

In this section, styrene is used to demonstrate the methodology.

## 4.4.1 Quality check of *passive\_sampling* data by comparing with EI data

The EI data for styrene shows a limited number of individual sources (22) in the Botlek/Pernis area, as illustrated in Figure 4.2.





The most important sources are concentrated in two areas, the so-called "sources 1 and 2". *Source 1* consists of ten separate sources emitting a total of 360 ton of styrene per year and *source 2* includes three sources close together emitting a total of 14 ton of styrene per year. Receptor locations of the passive samplers P13 and P14 are located near source 1, while receptor location P4 is located near source 2. Other styrene sources in the Botlek/Pernis area only contribute limited amounts to the styrene concentrations at the receptor sites and are therefore not included in the analysis.

Using the EI emission rates for source 1 and source 2, the actual meteorological conditions and the National Plume model, air concentrations of styrene at the fourteen receptor locations for the six periods have been calculated. These values are compared to the measured styrene concentrations (corrected for the background and traffic contributions, as described in section 4.3.1). The deviations are in general within a factor two, which is regarded a fair correlation.

### 4.4.2 Quality check of *active* sampling data by comparing with EI data

The styrene air concentrations have also been measured on a hourly basis at Maassluis and Hoogvliet. Therefore, it is possible to link air concentrations with wind directions during the sampling periods. From a first exercise in the analysis it was learned that the results improved when only air concentrations were used which were measured at wind speeds larger than 3 m/s. Analogous to the section above, EI data on the emission of styrene within the Botlek/Pernis area have been used to calculate the air concentrations of styrene at Maassluis and Hoogvliet. Figure 4.3a and 4.3b shows the measured and calculated concentrations of styrene for Hoogvliet and Maassluis, as a function of wind direction.



Figure 4.3a The measured and calculated concentrations of styrene at Hoogvliet as a function of the wind direction.



Figure 4.3b The measured and calculated concentrations of styrene at Maassluis as a function of the wind direction.

TNO-MEP - R 97/435

42 van 87

Figures 4.3a and 4.3b illustrate that in Hoogvliet and Maassluis the styrene air concentrations are increased in the down-wind sectors of the Botlek/Pernis area respectively in the sectors 270-90 and 90-180. It is also noted that in both locations only about 25% of the air concentrations of styrene are above the detection limit of  $1 \mu g/m^3$ . In Hoogvliet (Figure 4.3a), both the measured and model calculations indicate two styrene emission sources: one at north/north-east (windsector: 0-45) and the other at north/north-west (windsector: 270-315). In Maassluis (Figure 4.3b) both the measured and model calculations (though with less agreement than in Hoogvliet) indicate a styrene source south/south-east (windsector: 120-180).

In view of the locations of the two styrene emission sources and their emission rate, as shown in Figure 4.2, and the orientation and distances of Hoogvliet and Maassluis to both these styrene sources, it is concluded that the EI data in combination with model calculations agree with the hourly active measurements.

## 4.4.3 Estimate of styrene emission rates using spatial linear regression and *passive* measurements

The methodology developed in this project to determine styrene emission rates of the two sources 1 and 2 from air measurements with *passive* samplers is demonstrated as follows. Spatial linear regression has been applied for each of the six sampling periods using background and traffic corrected values. This results in multiplication factors to modify the EI default emission strengths of the aforementioned source 1 and source 2. The regression is adequate as the number of receptor locations is larger than the number of sources, respectively, 14 and 2. Figure 4.4 shows the result for period 2 with a correlation coefficient of 0.98.



Figure 4.4 Measured and calculated concentrations of styrene for period 2 at fourteen receptor locations.

Figure 4.4 shows that the agreement of the passive sampling results at locations P1 to P14 with the calculated results is quiet good. The optimal fit with a correlation coefficient of 0.98 in Figure 4.4 was achieved with multiplication factors of 0.19 and 2.72 of the EI "default" emission rates for, respectively, source 1 and 2.

In Table 4.3, the results of the calculated multiplication factors for the EI emission rates of both the styrene sources 1 and 2 during the six periods are shown. Also, the correlation coefficients for the six fits are presented to illustrate the "goodness of fit" for each of the six periods.

Table 4.3Results of spatial regression calculation of styrene using two sources 1 and<br/>2 and the measured air concentrations at fourteen receptor locations. The<br/>multiplication factors for the EI emission rates are shown, and the quality<br/>of the fit (the correlation coefficient).

	Source 1	error source 1	source 2	error source 2	correlation coefficient
period 1	0.47	0.11	0.25	0.74	0.46
period 2	0.19	0.01	2.72	1.52	0.98
period 3	0.06	0.01	0.74	0.71	0.59
period 4	0.24	0.05	2.86	1.02	0.61
period 5	0.42	0.03	1.42	1.66	0.93
period 6	0.05	0.01	0.88	0.71	0.71

From the results presented in Table 4.3 it is concluded that:

- Source 1 appears to be overestimated by the EI emission rates by a factor of 2 (period 1 and 5) to 20 (period 6). The emission of source 2 is comparable to the EI emission rates within a factor four overestimation (period 1) to a factor 3 underestimation (period 4). The results indicate that styrene emissions and consequently air concentrations at ground-level fluctuate over the period of six two-weekly measurements from a factor 2 20 for source 1 and 0.3 3 for source 2 as compared to the EI emission rates;
- The relative error in the calculation of the multiplication factors of source 2 is larger than for source 1. This is largely due to the lesser quality of the fit for receptor locations close to source 2. The quality of the fits, indicated by the correlation coefficient in Table 4.3, is excellent for period 2 and 5 but worse for the other periods. It is concluded that receptor locations (to) close to the emission sources do not result in adequate fits.

## 4.4.4 Estimate of styrene emission rates using spatial regression and *active* measurements

Analogous to the section above, also with the hourly active measurements collected at Maassluis and Hoogvliet a spatial linear regression has been performed. The spatial regression results in multiplication factors of the emission rates of

44 van 87

sources 1 and 2 in the Botlek/Pernis area. The spatial regression results are presented in Table 4.4

Table4.4Results of the spatial regression calculation of styrene using two sources 1<br/>and 2 and the measured air concentration differences during relevant wind-<br/>sectors at Hoogvliet and Maassluis. The calculated multiplication factors<br/>for the EI emission rates are shown , and the quality of the fit (the correla-<br/>tion coefficient).

	Multiplication factor	correlation coefficient
Source 1	0.1 - 0.2	0.41
Source 2	4-6	0.85

Analogous to the passive sampling results, also the active sampling results indicate that the emission rate of source 1 is *overestimated* by the EI with a factor 5 - 10, while source 2 seems to be *underestimated* by the EI with a factor 4 - 6.

### 4.4.5 Discussion and conclusions

The results of styrene demonstrate that with prior information on emission rates and locations ( the so-called "default" EI values), a sufficient number of passive sampling locations and application of spatial linear regression, the method provides information on *temporal fluctuations* of emission rates (in accordance to the sampling period) and the *reliability* of calculated emission rates. This type of information is not available from EI emission data based upon *annual* emission rates.

Comparing the results of the hourly measurements in Table 4.4 with the results of passive sampling in Table 4.3 demonstrates that both the methods give comparable values for the multiplication factors of the EI emission rates. Considering the facts that the two methods use different measuring techniques (passive versus active sampling), different receptor locations (fourteen locations within the Botlek/Pernis area versus two locations outside the Botlek/Pernis area) and different dispersion calculations (two-weekly averaged versus hourly meteorological conditions) this agreement in the results provides confidence in the developed methodology.

## 4.5 Benzene

This section demonstrates the methodology in the case of benzene emissions from the Botlek/Pernis area.

## 4.5.1 Quality check of *passive* sampling data by comparison with EI data

The EI lists 184 individual sources of benzene in the Rijnmond area. The most important sources are shown in Figure 4.5. The size of the symbols is linear to the square root of the annual emission rate of benzene.



Figure 4.5 Benzene emission sources in the Rijnmond area according to the EI.

The strongest source (at x-coordinate: 800 and y-coordinate: 433) has an emission rate of almost 40 ton benzene per year. Major industrial benzene sources include refineries, chemical industry and storage tanks. From the EI it is known that the 22 strongest emission sources all emit benzene at a low height (storage tanks), while the two elevated sources rank 20 and 26 with heights of respectively, 213 and 40 m.

Figure 4.6 shows the strongest 27 sources of benzene in the Botlek/Pernis area by their ranking number: number 1 is the strongest source, while number 27 is the weakest.



Figure 4.6 Benzene emission sources in the Botlek/Pernis area according to their EI emission rate (1-27) and the location of the fourteen receptor locations (P1-P14).

For each of the six sampling periods, the benzene air concentrations have been calculated at each of the fourteen receptor points using the 27 benzene emission sources, actual meteorological conditions and the National Plume model. As an example, Figure 4.7 shows for period 2 at each receptor location the calculated and measured air concentration of benzene. The measured concentrations have been corrected for background and traffic contribution, as described in section 4.3.1



Figure 4.7 Benzene air concentrations calculated and measured values at the fourteen receptor locations in the Botlek/Pernis area in period 2.

46 van 87

(Note that in Figure 4.7, the air concentration of benzene in period 2 at location 8 is missing, due to loss of the passive sampler). Figure 4.7 illustrates that measured and calculated concentrations for period 2 agree reasonably well, with

a few exceptions. Similar to period 2 also the results of other periods have been compared with EI data. From this comparison, it is concluded that passive sampling results agree reasonable with EI calculated values during the three-monthly measuring campaign.

## 4.5.2 Quality check of *active* sampling data by comparing with EI data

Analogous to the styrene analysis, also benzene concentrations measured by active measurements at Hoogvliet, Maassluis and Zwartewaal have been analysed. In line with the discussion in the section on passive sampling, only the 27 strongest sources in the Botlek/Pernis area have been used to calculate the air concentrations at all hours of the measuring period in Hoogvliet, Maassluis and Zwartewaal.

At Hoogvliet, Maassluis and Zwartewaal, the air concentrations of benzene in the wind sectors representing transport of air pollution from the Botlek/Pernis area to the three monitoring locations are elevated both for measured and EIcalculated values. It is concluded that despite differences in the level of air concentrations of benzene, both the hourly measured and the EI calculated air concentrations agree in identification of benzene sources.

# 4.5.3 Estimate of benzene emission rates using Bayesian statistics and *passive* measurements

Analogous to the styrene exercise, the method is demonstrated to calculate emission rates of benzene sources from air concentration measurements at the fourteen receptor locations. Again, the EI emission rates are treated as "default" values and multiplication factors for the emission rates are calculated to fit the measurements. The number of unknowns (27), which equals the number of multiplication factors for the 27 emission sources, *exceeds* the number of known parameters (14), which equals the number of measured benzene concentrations at fourteen receptor locations. Therefore, a spatial linear regression as performed for styrene in section 4.4.3 is not possible, and consequently, the Bayesian method referred to in section 3.2.4 has been applied. Figure 4.8 shows multiplication factors for the 27 source emission rates of benzene including the uncertainty as error bars for period 2.



Figure 4.8 The 27 multiplication factors for the EI emission rates of the 27 emission sources of benzene in period 2 calculated with Bayesian statistics.

Figure 4.8 illustrates that in period 2 the measurements only restrict four sources (1,2, 6 and 11) to values deviating significantly from unity. Of the other sources the measurements in this period contains little information, although the results suggests a lower emission rate than the EI default values. Similar to period 2 also for the other two-weekly periods, the multiplication factors of the EI default values have been calculated.

The results of the average emission rates in ton benzene per year for the 27 sources in the Botlek/Pernis area and the range of measured benzene air concentrations during the six periods at the fourteen passive locations, are presented in Figure 4.9.

#### TNO-report

TNO-MEP - R 97/435



Figure 4.9 Benzene emission rates in ton/year of the emission sources in the Botlek/Pernis area calculated from the results of passive sampling, as indicated by the range of benzene air concentrations  $(\mu g/m^3)$ .

In Table 4.5, the "default" EI emission rates and the multiplication factors of these rates as computed from the passive sampling results are presented. Also, information on the emission height, as retrieved from the EI database is provided.

TNO-MEP - R 97/435

#### 50 van 87

	EI	pass	sive	height	
source	Annual emission (ton/year)	fac	tor	(m)	
1	39	0	± 0	< 10	
2	18	0	± 0	< 10	
3	10	0.1	± 0.1	< 10	
4	10	0.1	± 0.2	< 10	
5	10	-	-	< 10	
6	9	0	± 0	< 10	
7	9	0	± 0	< 10	
8	8	0.6	± 0.4	< 10	
9	8	0	± 0	< 10	
10	7	0.5	± 0.4	< 10	
11	7	1.6	± 0.4	< 10	
12	5	1.9	± 0.7	< 10	
13	4	0.4	± 0.4	< 10	
14	4	0.3	± 0.4	< 10	
15	4	0	± 0	< 10	
16	4	0.9	± 0.4	< 10	
17	3	0.8	± 0.6	< 10	
18	3	0	± 0	< 10	
19	3	0.4	± 0.5	< 10	
20	3	-	-	213	
21	3	0	± 0	< 10	
22	3	-	-	< 10	
23	2	2	± 0.9	< 10	
24	2	-	_	< 10	
25	2	-	-	< 10	
26	2	_	-	40	
27	2	0	± 0.1	< 10	

 Table 4.5
 The height of emission sources, the EI emission rates of benzene in ton/year and multiplication factors of the EI emission rates computed from passive sampling results.

It is noted in Table 4.5, that for a number of sources the multiplication factor for the emission rate is calculated as zero values (sources:1, 2, 6, 7, 9, 15, 18 and 21). This result of the statistical analysis indicates that these sources are **not** contributing to the air concentrations of benzene at the fourteen receptor locations. For sources 5, 20, 22, 24-26, no multiplication factor is provided as the uncertainty in the values is too large. Hence, the sources are too far from sampling locations or too high to contribute to ground-level concentrations. Sources 16 and 17 show factors within an order of magnitude from unity, which indicates that the emission rates are *equal* to default EI values. Sources 11, 12 and 23 are *underestimated* by the EI, while the other sources 3, 4, 8, 10, 13, 14, 19 and 27 all show values less than unity, which is an indication of *overestimation* by the EI of actual benzene emissions.

It is concluded that the measured air concentrations of benzene by passive samplers *within* the industrial area provides a monitoring instrument to control benzene emissions at *two-weekly frequency*, especially by sources emitting at *low* heights: storage tanks and other diffuse emissions.

# 4.5.4 Estimate of benzene emission rates using Bayesian statistics and *active* measurements

Again analogous to the styrene exercise, the method is demonstrated to calculate emission rates of benzene sources from hourly air concentration measurements at Maassluis, Hoogvliet and Zwartewaal. The EI emission rates are treated as default values and "multiplication factors" for the emission rates are calculated to fit the measurements. Similar to the passive sampling results, spatial linear regression as performed for styrene in section 4.4.3 is not possible, and consequently, the Bayesian has been applied.

The results of the average emission rates in ton benzene per year for the 27 sources in the Botlek/Pernis area and the windroses at Maassluis, Hoogvliet and Zwartewaal of the average, hourly measured benzene air concentrations during the three-monthly are presented in Figure 4.10.



Figure 4.10 Benzene emission rates in ton/year of the emission sources in the Botlek/Pernis area calculated from the results of hourly measurements, as indicated by the windroses in Maassluis, Hoogyliet and Zwartewaal.

52 van 87

The windroses at Hoogvliet, Maassluis and Zwartewaal illustrate that air concentrations of benzene in the wind sectors representing transport of air pollution from the Botlek/Pernis area are elevated. In Table 4.6, the "default" EI emission rates and the multiplication factors of these rates as computed from the (background and traffic corrected) hourly measurement results are presented. Also, information on the emission height, as retrieved from the EI database is provided.

	EI	active factor		height
source	Annual emission (ton/year)			(m)
1	39	0	± 0	< 10
2	18	0	± 0	< 10
3	10	0	± 0	< 10
4	10	0	± 0	< 10
5	10	0.6	± 0.1	< 10
6	9	0	± 0	< 10
7	9	0	± 0	< 10
8	8	0.5	± 0.1	< 10
9	8	0.4	± 0.1	< 10
10	7	0.1	± 0	< 10
11	7	0.7	± 0.3	< 10
12	5	0	± 0	< 10
13	4	0	± 0	< 10
14	4	0	± 0	< 10
15	4	0	± 0	< 10
16	4	-	-	< 10
17	3	0	± 0	< 10
18	3	0	± 0.4	< 10
19	3	0	± 0	< 10
20	3	-	-	213
21	3	0	± 0	< 10
22	3	1.1	± 0.2	< 10
23	2	2.5	± 0.4	< 10
24	2	0	± 1.5	< 10
25	2	0	± 0	< 10
26	2	-	-	40
27	2	3.5	± 2.6	< 10

Table 4.5	The height of emission sources, the EI emission rates of benzene in ton/year
	and multiplication factors of the EI emission rates computed from hourly
	measurements.

receptor sites.

It is noted in Table 4.6, that for a number of sources the multiplication factor for the emission rate is calculated as zero values (sources: 1-4, 6, 7, 12-15, 17-19, 21 and 24-25). This result of the statistical analysis indicates that these sources are **not** contributing to the air concentrations of benzene at the three receptor locations. Source 22 shows factor within an order of magnitude from unity, which indicates that the emission rate is *equal* to default EI values. Sources 23 and 27 are *underestimated* by the EI, while the other sources 5 and 8-11 show values less than unity, which is an indication of *overestimation* by the EI of actual benzene emissions. Sources 16, 20 and 26 do not result in multiplication factor with adequate certainty. Hence, no information on the emission sources is collected by the measurements. It is assumed that emission by high stacks in between numerous low emissions can not be detected by the three

It is concluded that the measured air concentrations of benzene by hourly measurements *outside* the industrial area does not provide a monitoring instrument to control benzene emissions from on high stacks.

## 4.5.5 Comparison of benzene emission rates computed from *passive* and *active* sampling

In the former two sections, benzene emission rates of sources in the Botlek/Penis area have been computed both from two-weekly passive sampling and hourly active measurements. In this section, the results are compared to analyse the advantages of both methods. In Table 4.6, multiplication factors of the emission rates as assessed by both methods are summarised.

54 v	an	87
------	----	----

	EI	fac	tor	height	
source	Annual emission (ton/year)	active passive		(m)	
1	39	0.0	0.0	< 10	
2	18	0.0	0.0	< 10	
3	10	0.0	0.1	< 10	
4	10	0.0	0.1	< 10	
5	10	0.6	-	< 10	
6	9	0.0	0.0	< 10	
7	9	0.0	0.0	< 10	
8	8	0.5	0.6	< 10	
9	8	0.4	0.0	< 10	
10	7	0.1	0.5	< 10	
11	7	0.7	1.6	< 10	
12	5	0.0	1.9	< 10	
13	4	0.0	0.4	< 10	
14	4	0.0	0.3	< 10	
15	4	0.0	0.0	< 10	
16	4	-	0.9	< 10	
17	3	0.0	0.8	< 10	
18	3	0.0	0.0	< 10	
19	3	0.0	0.4	< 10	
20	3	-	-	213	
21	3	0.0	0.0	< 10	
22	3	1.1	1.1	< 10	
23	2	2.5	2	< 10	
24	2	0.0	-	< 10	
25	2	0.0	-	< 10	
26	2	-	-	40	
27	2	3.5	0.0	< 10	

Table 4.6The height of emission sources, the EI emission rates of benzene in ton/yearand multiplication factors of the EI emission rates computed from both passive sampling and hourly measurements.

From Table 4.6 it is learned that the triangle of three sampling locations at several kilometers from the industrial area does not monitor stack emissions among low sources, while passive samplers provide more adequately emissions from lower sources (lower than 10 m), including diffuse emissions. Based upon this information, a cost-effective abatement strategy may be developed to reduce benzene (and other hydrocarbon) emissions especially from low sources. In Figure 4.11a, b and c, the ground-level iso-concentration maps of benzene are presented, as a results of emission rates computed from respectively, the EI default values, passive sampling and hourly measurements.



Figure 4.11 Three monthly iso-concentration map of ground-level benzene in  $\mu g/m^3$  computed from EI default values (a), passive sampling results (b) and hourly measurements results<sup> $\circ$ </sup>.

TNO-MEP - R 97/435

56 van 87

Figure 4.11 illustrates that all three methods (EI, passive sampling and direct measurements) indicate that the highest benzene ground-level concentrations are around two main emission source areas: the first at x/y co-ordinates: 80000/433000 and the second at 85000/434000. Figure 4.11c also illustrates that the benzene emissions of the high stacks do *not* result in elevated ground-level concentrations. Hence, high stacks dilute effectively gaseous pollution.

In order to assess the cost-benefits of both passive sampling and hourly measurements, a first-order costs estimate is performed:

- *passive sampling*; In the three months period, six times fourteen passive sampling periods have been performed. This totals 90 analysis of passive sampling, including six duplicates. The cost of one analysis for benzene, including the costs of a sampler is 100 ECU. Hence, for analysis the costs are 9000 ECU. Staffing inputs for the measurements concern about seven field trips (one to select locations and six to change samplers), which costs in total about 8000 ECU. So, the total costs for sampling and analysis is 17000 ECU.
- hourly measurements; In the three monthly period, benzene measurements have been implemented by three so-called "BTX" monitors. The cost of a monitor is about 25000 ECU. The cost of weather-proof housing, including power supply, air-conditioning and gases is about 35000 ECU. Hence, in to-tal for three locations 1800000 ECU is required. A depreciation period of four years results in the costs of a three monthly campaign of about 15000 ECU. During this period, also seven field trips are envisaged to set-up and break-down the equipment and five inspection trips, which costs about 8000 ECU. So, the total costs for sampling and analysis is 23000 ECU.

This illustrates the cost-effectiveness of passive sampling, but even more important are the flexibility of the locations of passive samplers over fixed weather-proof locations and lack of problems common in field campaigns such as power failure, equipment problems and the need for chemicals.

### 4.5.6 Discussion and conclusions

The results of benzene demonstrate that with prior information on emission rates and locations (the so-called default EI values), a sufficient number of passive sampling locations and a triangle of up- and downwind locations with hourly measurements and application of Baysian statistics, the method provides information on *temporal fluctuations* of emission rates (in accordance to the sampling period) and the *reliability* of calculated emission rates. Both passive and active sampling give comparable results, but passive sampling *within* the industrial area details especially low sources. Passive sampling is preferred as it is a cost-effective, flexible and robust method. It is noted that even in case of a large number of emission sources (27) and a limited number of passive sampling

locations (14), the method results in adequate emission rates with emphasis on low sources.

## 4.6 C<sub>2</sub>-C<sub>5</sub>

In the next section, the methodology for " $C_2$ - $C_5$ " hydrocarbons is demonstrated with emphasis on ethene.

### 4.6.1 Factor analysis

The " $C_2$ - $C_5$ " concentrations at Zwartewaal have been analysed with factor analysis in order to identify source categories contributing to the air concentrations. Table 4.5 shows the results of the factor analysis. The output of factor analysis are loadings of each compound on a number of factors, as well as the correlation coefficients between the factors. A total of three factors explained more than 80% of the total variance in the data.

- Marine and Constra	factor I	factor II	factor III
acetylene			0.856
ethane		0.624	
ethene		0.914	
i-butane	0.827		
i-pentane	0.935		
n-butane	0.876		
n-pentane	0.970		
propane	0.408	0.768	
propene		0.983	
Factor correlation			
factor I	1	0.334	0.311
factor II	0.334	1	0.413
factor III	0.311	0.413	1

Table 4.7 Results of factor analysis on " $C_2$ - $C_5$ " concentrations measured at Zwartewaal. Only factor loadings are presented which exceed the value of 0.4.

Table 4.7 shows that factors I, II and III correlate only weakly (0.31 - 0.41) indicating an adequate distinction between source categories in the data. Factor I is characterised by high loadings of i/n-butane and i/n-pentane which is an indication for evaporation from storage tanks. Factor II shows high loadings of ethane, ethene, propane and propene, which is an indication for chemical industries in general. Factor III only presents acetylene which is an indication for traffic emissions.

58 van 87

### Factor I: evaporation from storage tanks

Factor analysis shows that the four components i/n-butane and i/n-pentane are measured in a fixed profile. From EI data it is known that these four compounds figure prominently in the emission profile of storage tanks in the Bot-lek/Pernis area. In Figure 4.12 the measured concentrations of i/n-butane and i/n-pentane in Zwartewaal, as a function of winddirection are presented. This includes only hours for which the windspeed is larger than 3 m/s.



Figure 4.12 Hourly i/n-butane and i/n-pentane air concentration in  $\mu g/m3$  in Zwartewaal in different windsectors and during windspeeds larger than 3 m/s.

Figure 4.12 shows two prominent peaks of i/n-butane and i/n-pentane air concentrations. High levels of i/n-butane and i/n-pentane are measured at eastern wind directions (windsector: 60-120). This points at a number of storage tanks at a distance between 2.5 and 5 km easterly from Zwartewaal *inside the Botlek/Pernis area*. High levels of i/n-butane and i/n-pentane air concentrations at northern wind directions (windsector: 300-10) identifies storage tanks between Zwartewaal and Maassluis at a distance of 3 km *outside the Botlek/Pernis area*.

There is a slight difference between the profiles from the two sectors: in the profile from the east i-pentane represents the largest fraction, whilst in the northern profile n-butane dominates. The sector-averaged profiles are shown in Table 4.8 together with the profile of gasoline evaporation from the literature.

Table 4.8Average measured profiles for i/n-butane and i/n-pentane measured atZwartewaal in easterly and northern windsectors. Also shown is the profileof gasoline evaporation emitted by storage tanks from literature.

and the second second	i-butane	i-pentane	n-butane	n-pentane
sector north:	0.13	0.28	0.37	0.22
sector east:	0.15	0.35	0.32	0.18
gasoline evaporation:	0.14	0.36	0.29	0.21

Table 4.8 shows that the profile from the east agrees reasonably well with the gasoline evaporation profile, while the northern profile deviates from the latter profile. A possible explanation for the deviating profile of the emission from the storage tanks north of Zwartewaal may be the storage of for example kerosine (instead of gasoline).

Factor I shows that the method is an excellent instrument to identify and monitor emissions from storage tanks by measuring up-and down-wind VOC air concentrations at hourly sampling periods.

#### Factor II: chemical industry

Factor II is characterised mainly by the compounds ethene and propene and less by the compounds ethane and propane. Especially, ethene and propene air concentrations are well correlated with a coefficient of correlation of 0.87 and a ratio of ethene to propene air concentrations of 6. Hence, factor II is well represented by ethene air concentrations, which indicates refinery processes and chemical base industries in general. In Figure 4.13 the air concentrations of ethene in Zwartewaal are presented. These air concentrations are corrected by the background and traffic emissions using the Maassluis air concentrations at relevant winddirections. An overall traffic contribution at Zwartewaal has been estimated using the traffic profile. This contribution is found to be marginal compared to the measured levels used in the analysis below.







Figure 4.13 Measured air concentrations of ethene at Zwartewaal as a function of winddirection, including only hours for which windspeeds exceed 3 m/s.

Figure 4.13 shows that ethene air concentrations strongly peaks during northeasterly winds (windsector: 30-75). This indicates a (or a cluster of) ethene source(s) north-east from Zwartewaal. Taking into account, the measured ethene concentrations at Maasluis and Hoogyliet during the relevant winddirections (south-east for Maassluis and north-west for Hoogvliet) limits the distance of the ethene source to less than 4 km north-east from Zwartewaal. A source at a larger distance would result in higher ethene concentrations, as calculated with the dispersion model, than the ethene concentrations actual measured at Hoogvliet and Maassluis.

Inspection of provisional data of the Emission Inventory of 1996 results in a potential ethene source north-east of Zwartewaal. The source is a chemical plant emitting ethene from a low height. The emission is distributed over the whole terrain of 250x250 m and yields an annual total of 7923 kg. Figure 4.14 shows the measured and calculated ("model") ethene concentrations at Zwartewaal for all hours for which ethene has been measured in the wind directions from 0 - 90 degrees.



Figure 4.14 Model and measured ethene concentrations in Zwartewaal in de windsector 0-90.

Figure 4.14 illustrates that the *calculated* concentrations based upon the default EI data are a factor 10 to 100 times smaller than the *measured* concentrations. Note the difference of a factor ten between the two vertical axes for the model calculations and the measurements! Possible explanations for the differences between the measurements and the model are:

- the dispersion model underestimates the concentrations from a surface source;
- 2. the emission rate from the Emission Inventory is too low; and/or
- 3. an unknown ethene source.

In view of the agreement between measurements and model calculations for benzene, which was emitted mostly by surface sources, the first explanation is not likely. This leads us to the conclusion that the emission rate of the ethene source is a factor 10 to 100 times larger than the EI value or an unknown ethene source has been identified. Further information on possible new ethene related activities is required to further investigate these ethene emissions. This is beyond the scope of this project.

The example of ethene demonstrates that up- and down-wind monitoring and simple spatial regression statistics may help to identify (unknown) VOC emissions.

61 of 87

62 van 87

#### Factor III: traffic emissions

Factor III is characterised by a high loading of acetylene. A plot of the air concentrations of acetylene in Zwartewaal as a function of the windsectors shows that acetylene is also elevated in the same windsector as ethene. This indicates that the identified ethene source is also emitting acetylene. However, the temporal behaviour of acetylene is different from ethene which indicates the influence of another source (otherwise acetylene would have a high loading on factor II). Considering the relatively large contribution of traffic (73%) to the total emission of acetylene, as indicated in Table 3.1, suggests that traffic is the most likely source. Traffic emissions are beyond the scope of this research and therefore acetylene is not further elaborated.

#### Discussion and conclusions

The example of " $C_2$ - $C_5$ " analysis illustrates the application of factor analysis to identify various *source categories* emitting specific profiles of " $C_2$ - $C_5$ ". Plotting air concentrations of " $C_2$ - $C_5$ " components which are representative for the emission profiles, as a function of the wind sectors provides information on the location of the source categories. Using EI information and air concentrations of other receptor locations in relevant wind sectors, identifies the location of the potential sources. Finally, model calculations using actual meteorological conditions and a fit with the measured air concentrations provides emission rates of specific components. A large ethene source was identified following this method which was (in 1995) not registered in the EI. Further investigation is beyond the scope of this project.

### 4.6.2 C<sub>2</sub>-C<sub>5</sub> emission rates by FTIR measurements

The results of the VOC measurements with emphasis on " $C_2$ - $C_5$  VOC" air concentration measurements obtained by FTIR remote sensing during a week within the Botlek/Pernis area are evaluated in this section. The FTIR results are presented in Table 4.9.

beam	μ <b>g/m</b> ³					
position	Ethane	Propane	Butane	Pentane		
1						
2						
3						
4						
5						
6						
7			31	37		
8	11	32	540			
9	35	129	248			
10			28			
11						
12				203		
13				160		
14		13				
15						
16						
17		7				
18			47			
19			188			
20			83			
21			859			

Table 4.9The air concentrations on each beam position of the FTIR in the Bot-<br/>lek/Pernis area.

The beam positions in Table 4.9 in accordance to the four regions a to D within the Botlek/Pernis area are indicated in Figure 3.4. In addition to " $C_2$ - $C_5$  hydrocarbons", also methane, benzene and hydrocarbons higher than  $C_6$  were measured, but the results have not been indicated in Table 4.9. The detection limit of FTIR for individual compounds in ambient air is about 2 mg/m<sup>3</sup> per meter and hence for the 200-300 meter path length applied during this campaign, the detection limit is about 5-10 µg/m<sup>3</sup> per " $C_2$ - $C_5$  hydrocarbon". As illustrated in Table 4.9 for most " $C_2$ - $C_5$  VOC", the ground-level air concentrations within the Botlek/Pernis area are below these values, but especially the beam positions 7-10 in the region B (see Figure 3.4) indicate elevated ground-level " $C_2$ - $C_5$  VOC" air concentrations. In the following two sections, the results for air concentrations of benzene (region A) and butane (region B) are used to illustrate the FTIR methodology to estimate VOC emissions.

TNO-MEP - R 97/435

### benzene by FTIR

In region A at beam position 2 (see Figure 3.4), elevated ground-level benzene concentrations were measured due to benzene emissions up-wind (south) of the beam. The results are shown in Figure 4.15.



Figure 4.15 A time series of benzene concentrations measured at beam position 2 in region A of the Botlek/Pernis area.

Figure 4.15 shows that two peak concentrations of benzene occurred with a duration of about 200 seconds for each peak with a maximum concentration of about 110  $\mu$ g/m<sup>3</sup>. There was no evidence of wind direction change during the measurements which suggests that a temporal change of emission rates is the cause of the variation in benzene air concentrations. In Figure 4.16, the benzene emissions as a function of height of the benzene emission sources are presented.



Figure 4.16 Benzene emission at 1.5 m, 6.0 m and 15.0 m height in region A of the Botlek/Pernis area in ton per year (tpa) and the maximum peak during the measuring campaign in ton per year.

66 van 87

TNO-MEP - R 97/435

The benzene emissions up-wind of the beam position was calculated in the grid 640 - 880 (x-axe) and 3120 - 3360 (y-axe). Especially, in the grid cell 800 (x-axe)/ 3360 (y-axe), a benzene emission source of more than 5 ton per year is identified. Site inspection and Emission Registration provides information on storage tanks in the aforementioned grid. The FTIR results suggests that a single tank is responsible for the benzene emissions measured at beam 2.

### $C_2$ - $C_5$ by FTIR

The case of butane measurements in region B at beam positions 7-11 (see Figure 3.4), is used to illustrate the potential of FTIR to establish  $C_2$ - $C_5$  emission rates. In Figure 4.17 the calculated butane emissions in region B are shown as a function of the height of the butane emission source.



Figure 4.17 Butane emission at 1.5 m, 6.0 m and 15.0 m height in region B of the Botlek/Pernis area in ton per year (tpa) and the maximum peak during the measuring campaign in ton per year.

67 of 87

68 van 87

Figure 4.17 presents butane emissions up-wind of the beam positions 7-11 in the grid 2900 - 3500 (x-axe) and 1500 - 2700 (y-axe). The major source of butane emissions have been located between beams 8 and 9, on the premises of a chemical factory. The butane emissions in region B are not correlated with pentane emissions, which suggests that butane is emitted by a chemical process and not by evaporation of a storage tank of hydrocarbons. Similar to butane and benzene also for methane, ethane, propane, pentane and the sum of "C6 and higher VOC" the emission rates in the various regions A to D (see Figure 3.4) have been calculated.

Table 4.10 Emission rates of " $C_1$ - $C_6$  and higher VOC" in the regions A to D in the Botlek/Pernis area.

Region	Emission Rate ( ton per year)					
	Methane	Ethane	Propane	Butane	Pentane	C6 and higher
А	1100					820
В	1670	210	270	940	160	2240
С	1620					1030
D	300			70		1320

It is noted in Table 4.10 that methane has been included for information purposes as this component has only been measured by FTIR and not by the other methods in this research. The methane measurements have been corrected for "background" methane, which is about 1.2 mg/m<sup>3</sup>, hence the emission rates in Table 4.10 are related to Botlek/Pernis emissions (leakage of natural gas?). The ER emission rates of the whole of the Botlek/Pernis area for propane and butane are respectively 320 and 2350 ton per year, which agrees reasonable with the 270 and 940 in the B-region measured with FTIR. It is noted that the sum of the emissions of methane and "C<sub>6</sub> and higher VOC" in region A to D are respectively 4700 and 5410 ton per year.

#### Discussion and conclusions

FTIR has previously mainly applied to assess VOC emissions from individual plants, where beam positions provide overlapping data and ground-level concentrations of VOC are higher than in the Botlek/Pernis area. Under the conditions of the demonstration in the Botlek/Pernis area, the degree of overlap of the beams was minimal and therefore, the results could not illustrate the full potential of the method. However, it is concluded from the results that FTIR is very well suited to identify VOC emissions from a single tank and to provide information on emission rates of methane and the sum of "C<sub>6</sub> and higher VOC" which was not available from the other methods applied in this research.

#### 4.6.3 C<sub>2</sub>-C<sub>5</sub> air concentrations by HAWK measurements

 $C_2$ - $C_5$  air concentrations were also measured by two HAWK remote sensing monitors for a period of two weeks at Zwartewaal and Maassluis. The results of the HAWK measurements showed a diurnal pattern with in- and decreasing air concentrations during respectively night and day.

The detection limit of individual " $C_2$ - $C_5$  VOC" by HAWK is about 5 µg/m<sup>3</sup> and the relative response for methane, ethane, n-butane, n-pentane and n-hexane is respectively 0.06, 0.6, 1.2, 1.5 and 1.5. The background concentration of methane is about 1200 µg/m<sup>3</sup>, while the sum of " $C_2$ - $C_5$  VOC"air concentrations (as indicated in Annex 1) at Maassluis and Zwartewaal measured by other methods in this project are about 50 µg/m<sup>3</sup>. Hence, the HAWK detection is dominated by ambient methane air concentrations.

From these results and the specifications of the HAWKS, it is concluded that the measured fluctuations at Maassluis and Zwartewaal reflect more a less the natural diurnal fluctuation of the background concentration of methane due to the de- and increasing mixing height. Hence, no VOC emissions from the Bot-lek/Pernis area could be established by the HAWK measurements. It is recommended to limit VOC emission measurements with HAWK remote sensing equipment to plant conditions, where the ratio methane/C<sub>2</sub>-C<sub>5</sub> is much lower than under ambient air conditions.

70 van 87

TNO-MEP - R 97/435

## 5. Conclusions and Recommendations

#### findings

The demonstrated methodology to assess " $C_2$ - $C_8$  hydrocarbon" emissions from an industrial area is summarised by the following steps: i.) collection of prior information on emission-rates and -locations (i.e. from an Emissions Inventory database or production characteristics in an industrial area), ii.) collection of hourly-averaged meteorological data, iii.) two-weekly passive sampling *within* an industrial area for " $C_6$  and higher hydrocarbons" and/or hourly measurements *outside* an industrial area for " $C_2$ - $C_5$  hydrocarbons" and " $C_6$  and higher hydrocarbons", and iv.) data analysis by spatial linear regression and Baysian statistics in combination with a short-term inverse-dispersion model.

The results of data analysis of the measuring campaign in and around the Botlek/Pernis area indicate that *styrene* emission rates of the most important sources are a factor 10 smaller than the values in the EI database of 1995; *benzene* emission rates of the 27 most important sources agree within a factor 2-4 with the EI database, with the exception of the stack sources at 40 and 213 m; and an unknown *ethene* emission source in the Botlek/Pernis area was identified. The measured two-weekly averaged emission rates of styrene and benzene show fluctuations over an order of magnitude and consequently, also groundlevel air concentrations of these compounds.

#### reliability

The *reliability* to determine VOC emission rates by combined measurements and modelling depends on a number of factors:

- it is required that meteorological conditions during the period of air sampling are adequately determined. Atmospheric stability and wind direction are major factors contributing to the dispersion and transport of air pollutants. In this project, general meteorological conditions were measured by a meteorological station at a vicinity of 30 km from the Botlek/Pernis area, while wind direction was measured in the centre of the Botlek/Pernis area. This approach assured a relatively adequate estimate of hourly-averaged meteorological conditions;
- a second aspect is the quality of the atmospheric dispersion model. It is important to use a model in accordance to the conditions characterised by parameters as follows: i.) distance of the emission source to the sampling location, ii.) hourly-averaged air concentrations and meteorological data and iii.) information from the EI on the emission source characteristics such as stack height and plume rise. The updated National Plume model applied in this project is the "state-of -the-art" Gaussian plume model in the Netherlands for hourly dispersion modelling;
- a third aspect is the linear relation between emission and ground-level concentration of the VOC compounds, i.e. when the effects on the concentra-

72 van 87

tion of deposition and chemical conversion during transport from the source to the monitoring location are small. The selected " $C_2$ - $C_8$  hydrocarbons" in this project comply with this condition; and

• a fourth aspect are the uncertainties in comparing EI emission rates with the computed data. The EI data are based upon *annual* emissions from 1995, while emissions calculated in the project are based upon a three-monthly campaign in 1997. In addition, the EI data address emissions based upon "routine" industrial activities, while the measurements monitor VOC emissions during actual production conditions. Based upon these considerations, it is concluded that comparison of calculated emission rates by the proposed methodology and the EI emission rates only provides a "first order" estimate about the reliability of the method.

#### conclusions

Taking into account the aforementioned uncertainties, it is concluded that the results presented in this report for styrene, benzene and ethene demonstrate that the method provides adequate information on the *spatial distribution* of emission sources, *chemical profiles* of emission sources, *temporal fluctuations* of emission rates, *distinction in low (diffuse) and high (stack)* emissions and the *reliability* of the calculated emission rates.

A second conclusion is that the most cost-effective and simple methodology which can be applied to assess emission rates of " $C_6$  and higher VOC" (including chlorinated compounds) by *low* sources is based upon passive sampling within the industrial area. For " $C_2$ - $C_5$  VOC" emissions, a triangle of upand downwind receptor locations are required based upon hourly on-line measurements and/or active sampling.

A third conclusion is that the developed method provides an important instrument for enforcement agencies to monitor actual VOC emissions from a complex industrial sources, not only on an annual but even at a two-weekly basis. This is an important aspect in order to check the effects of measures to reduce industrial VOC emissions.

A fourth conclusion concerns VOC emission measurements by the "HAWK" and "FTIR" remote sensing techniques. It is concluded that the HAWK is not an adequate technique to estimate VOC emissions from an industrial area, due to its lack of sensitivity for " $C_2$ - $C_5$  hydrocarbons" as compared to methane air concentrations. Under plant conditions with a more favourable " $C_2$ - $C_5$  hydrocarbons"/methane ratio, HAWK measurements is a suitable on-line detection technique, due to its simple operation as compared to automated GC. The FTIR in combination with Bayesian statistics is a promising method to determine VOC emission rates, though (due to the relatively high detection limits) the method is more adequate for on plant measurements.
Finally, it is concluded that the objectives of the project: to develop and demonstrate a methodology to assess VOC emissions, and to estimate the contributions by various source categories in a complex industrial area based upon ground-level VOC measurements have been achieved. Presently, there is no need for further development of the method, but there is definitely need for more practical experiences.

#### recommendations

It is recommended to apply the method for at least over a period of a year and to compare the results with EI emission data. Furthermore, it is recommended to apply the method for an industrial area without Emission Inventory data and finally, to apply the method within an industrial complex to monitor diffuse emissions and (if possible) in combination with the FTIR method.

74 van 87

TNO-MEP - R 97/435

## 6. Dissemination activities

Two workshops have been organised in the framework of the project, the first at March 6<sup>th</sup> and the second at December 4<sup>th</sup> 1997 at TNO in Apeldoorn, the Netherlands. During the workshops, the measuring and modelling strategies as well as the results were presented to national and international experts.

In addition to both the workshops, also, at a GENEMIS meeting at October 9<sup>th</sup> 1997 in Stuttgart in the framework of the EUROTRAC programme, the project was presented to a group of European senior scientists.

In 1998, publication in environmental magazines and a scientific journal is planned to further disseminate the developed methodology. In addition, an information sheet is prepared to be submitted to relevant authorities, industrial associations and private companies.

76 van 87

TNO-MEP - R 97/435

## 7. References

Bailey, J.C., K. Gunary, B. Schmidl and M.L. Williams (1990).	
Speciated hydrocarbon emissions from a sample of UK vehicles	on the road
over a range of speeds. Sci.Tot.Environ., 93, pp.199-206.	

Bailey, J.C. and Parkes (1993). Speciated hydrocarbons from the UK large scale survey. WSL LR 979, Stevenage.

Barrefors, G. and G. Petersson (1992).Volatile hazardous hydrocarbons in a Scandinavian urban road tunnel. Chemosphere, 25, pp.691–696.

Baugues, K. (1986).

A review of NMOC, NO<sub>x</sub> and NMOC/NO<sub>x</sub> ratios measured in 1984 and 1985.

EPA-450/4-86-015, U.S. Environmental Protection Agency, Research Triangle Park, NC; NTIS Document PB87166963.

Berdowski, J., M. Pulles and M. Roemer (1992) Emissies in het Rijnmondgebied. TNO rapport R92/024.

Berdowski, J. and J. Jonker (1994).

Emissies in Nederland – 1992; bedrijfsgroepen, regio's en individuele stoffen; ramingen 1993. Publikatiereeks Emissieregistratie, nr. 21, Hoofdinspectie Milieuhygiëne, VROM, Den Haag.

Bishop, G.A. and D.H. Stedman (1989). Oxygenated fuels, a remote sensing evaluation. SAE Paper No. 891116.

Boersen, G.A.W., C. Huygen, W.A.S. Nijenhuis en M.P.J. Pulles (1995). Handleiding MILIS, versie 1.3, TNO MEP, Delft, december 1995

BUA Bern (1986).

Bundesamt für Umweltschutz, Bern. Schadstoffemissionen des privaten Straßenverkehrs. Schriftenreihe Umweltschutz nr. 55, Bern.

Bullin, J.A., N.J. Green and J.C. Polasek (1980). Determination of vehicle emission rates from roadways by mass balance techniques. Environ.Sci.Technol., 14, pp.700–705.

Bullin, J.A. and J.C. Polasek (1981). Editorial Correspondence. Environ.Sci.Technol., 15, pp.365–366.

78 van 87

Cadle, S.H., R.A. Gorse and D.R. Lawson (1993).

Real-world vehicle emissions: a summary of the third annual CRC-APRAC on-road vehicle emissions workshop. J.Air&Waste, 43, pp.1084–1090.

CBS (1991).

Luchtverontreiniging, emissies door wegverkeer; parkemissies 1970–1990. Centraal Bureau voor de Statistiek, Voorburg.

Ching, J.K. (1994).

Rationale, experimental design and methodology for determining the overall accuracy of whole city VOC and  $NO_x$  inventories. Presented at the fourth US/German-European union workshop on photochemical ozone problem and its control in urban, regional and global scales, June 13–17, 1994, Charleston, SC.

Ching, J.K., J.H. Novak, K.L. Schere and N.V. Gillani (1987). Reconciling urban VOC/NO<sub>x</sub> emission inventories with ambient concentration data. Presented at the 80<sup>th</sup> annual meeting of APCA, June 21–26, 1987, New York.

Dannecker, W., B. Schröder and H. Stechmann (1990). Organic and inorganic substances in highway tunnel exhaust air. Sci.Tot.Environ., 93, pp.293–300.

Davis, J.C. (1973).

Statistics and data analysis in geology, second edition, John Wiley & Sons, New York

DCMR (1973).

Urgentieprogramma Sanering Rijnmondgebied; een eerste rapportering. Technische Commissie, Schiedam.

DCMR (1983).

Dossier Reactieve Koolwaterstoffen, Schiedam.

#### DCMR (1984).

Evaluatie van de koolwaterstofmetingen in de periode 1973–1980, Schiedam.

Derwent, R.G., D.R. Middleton, R.A. Field, M.E. Goldstone, J.N. Lester and R. Perry (1994).

Analysis and interpretation of air quality data from an urban roadside location in central London over the period from July 1991 to July 1992. Submitted to Atmospheric Environment.

Duuren, H. van, A.J. Elshout, F.G. Römer, J.W. Viljeer and E. Noks (1979). Luchtverontreiniging boven de Rijnmond, Elektrotechniek 57, 4, 267–277.

Eggleston, H.S. (1993).

Uncertainties in the estimates of emissions of VOCs from motor cars. Proceedings of the TNO/EURASAPworkshop on the reliability of VOC emission data bases (eds: Baars, Builtjes, Pulles, Veldt), 9–10 June 1993, Delft.

Feigley, C.E. and H.E. Jeffries (1979).

Analysisof processes affecting oxidant and precursors in the Los Angeles Reactive Pollutant Program (LARPP), Operation 33. Atmos.Environ., 13, pp.1369–1384.

Fujita, E.M. and B.E. Croes (1990).

Formaldehyde emissions in the South Coast air basin derived from ambient measurements, Paper presented at the A&WMA International Specialty Conference on Tropspheric Ozone and the Environment, Los Angeles, 22 March 1990.

Fujita, E.M., B.E. Croes, C.L. Bennett, D.R. Lawson, F.W. Lurmann and H.H. Main (1992).

Comparison of emission inventory and ambient conetration ratios of CO, NMOG, and NO<sub>x</sub> in California's South Coast Air Basin. J.Air Waste Manage.Assoc., 42, pp.264–276.

Fujita (1993).

Trends in emissions and ambient concentrations of CO, NMHC and  $NO_x$  in the South Coast Air Basin.

Proceedings: Southern California Air Quality Study.

Data Analysis ASWMA Specialty Conference, VIP-26, Air & Waste Management Association, Pittsburgh, PA.

Fujita, E.M. and D.R. Lawson (1994).

Evaluation of the emission inventory in the South Coast Air Basin. Desert Research Institute, Reno, NV.

Fujita, E.M, Watson J.G., Chow J.C (1995).

Receptor model and emission inventory source appointments of nonmethane organic gases in California' San Joaquin Valley and San Francisco Bay area., Atmospheric Environment 29, 3019-3035.

80 van 87

Gregori, M., C. Lanzerstorfer, H. Oberlinninger, H. Puxbaum, P. Biebl, O.
Gläser and J. Villinger (1989).
Tauerntunnel-Luftschadstoffuntersuchung 1988; Ergebnisse der Meßkam-

pagne vom 16-18 Sept. 1988. Bericht 4/89, Technische Universität Wien, Wien.

Gorse Jr., R.A., J.M. Norbeck (1981). CO emission rates for in-use gasoline and diesel vehicles. JAPCA, 31, pp.1094–1096.

Gorse Jr., R.A. (1984).
On-road emission rates of carbon monoxide, nitrogen oxides and gaseous hydrocarbons.
Environ. Sci. Technol. 18 (7): 500-507.

- Hampton, C.V., W.R. Pierson, D. Schuetzle and T.M. Harvey. (1983). Hydrocarbons gases emitted from vehicles on the road. 2. Determination of emission rates from diesel and spark-ignition vehicles. Environ. Sci. Technol. 17 (12): 699–708.
- Henry R.C., Lewis C.W., Hopke P.K., Williamson J.H. (1984). Atmospheric Environment 18,1507-1515

Heyes, C.J. and M.L. Williams (1993).

Assessment of the accuracy of the UK VOC-emissions. Proceedings of the TNO/EURASAP workshop on the reliability of VOC emission databases (eds: Baars, Builtjes, Pulles en Veldt), 9–10 June, 1993, Delft.

Hlavinka, M.W. and J.A. Bullin (1988).

Validation of mobile source emission estimates using mass balance techniques. JAPCA, 38, pp.1035–1039.

Huygen, C. (1995).

Het verband tussen  $NO_x$  emissies en  $NO_x$  concentraties in West-Nederland. TNO-rapport (in voorbereiding).

Ingalls, M.N. (1989).

On-road vehicle emission factors from measurements in a Los Angeles Area Tunnel. Paper No. 89-137.3, presented at the A&WMA 82<sup>nd</sup> Annual Meeting, Anaheim, CA.

Ingalls, M.N., L.R. Smith and R.E. Kirksey (1989).

Measurement of onroad vehicle emission factors in the California South Coast air basin-Volume I: regulated emissions. Report No. SwRI-1604 from Southwest Research Institute to the Coordinating Research Council, Atlanta, GA, June 1989. NTIS Document PB89220925.

Johansson, C. (1994).

Personal communication.

#### KEMA (1982).

Onderzoek betreffende luchtverontreiniging door stikstofoxiden (publicaties 1968–1981), 169–200, Arnhem.

Kennedy, G.J., J.T. White III, K.J. Springer and M.N. Ingalls (1975). Exhaust emissions from heavy duty trucks tested on a road course and by dynamometer. SAE Paper No. 750901.

#### Kuntasal, G. and T.Y. Chang. (1987).

Trends and relationships of  $O_3$ ,  $NO_x$  and HC in the South Coast air basin of California. JAPCA 37 (10: 1158–1163.

Lawson, D.R., P.J. Groblicki, D.H. Stedman, G.A. Bishop and P.L. Guenther (1990).Emissions from in-use motor vehicles in Los Angeles: a pilot study of re-

mote sensing and the inspection and maintenance program. J.Air Waste Manage.Assoc., 40, pp.1096–1105.

Lee, J.B, H. Westberg and B. Lamb (1988). NO<sub>x</sub> and VOC emission estimates derived from ambient data, Paper No. 88-66.5, presented at the APCA 81<sup>st</sup> Annual Meeting, Dallas, TX.

- Lightman, P., A.S. Kallend, A.R. Marsh, B.M. Jones and S.A. Penkett (1990). Seasonal variation of hydrocarbons in the free troposphere at mid-latitudes. Tellus 42B, pp.408–422.
- Lonneman, W.A., S.L. Kopczynski, P.E. Darley and F.D. Sutterfield (1974). Hydrocarbon composition of urban air pollution. Environ.Sci.Technol., 8, pp.229–236.

Lonneman, W.A., R.L. Seila eand S.A. Meeks (1986). Non-methane organic composition in the Lincoln tunnel. Environ.Sci.Technol., 20, pp.790–796. 82 van 87

Mayrsohn, H. and J.H. Crabtree (1976). Source reconciliation of atmospheric hydrocarbons. Atmos. Environ., 10, pp.137–143.

Morris, R.E. (1990).

Systems Applications, Inc., San Rafael, CA, personal communication., 1990; Final version: Protocol for modelling the air quality impact of fuel composition changes in light-duty vehicles, Report No. SYSAPP-90/056 from Systems

Applications, Inc., San Rafael, CA, to the Auto/Oil Air Quality Improvement Research Programme and to the Coordinating Research Council, Atlanta, GA, 9 July 1990.

Mulholland & Seinfield (1995).

Atmospheric Environment 29, 497-516, 1995

Mukund R., Kelly T.J., Spicer C.W. (1996).

Source attribution of ambient air toxic and other VOCs in Columbus, Ohio, Atmospheric Environment 30, 3457-3470

Nelson, P.F., S.M. Quigley and M.Y. Smith (1983).

Sources of atmospheric hydrocarbons in Sydney: a quantitative determination a using source reconciliation technique. Atmos.Environ., 17, pp.439– 449.

Penkett, S.A., N.J. Blake, P. Lightman, A.R. Marsh, P. Anwyl and G. Butcher (1993).

The seasonal variation of nonmethane hydrocarbons in the free troposphere over the North Atlantic Ocean: possible evidence for the extensive reaction of hydrocarbons with the nitrate radical. J.Geophys.Res., 98, pp.2865–2885.

Pierson, W.R., A.W. Gertler and R.L. Bradow (1990). Comparison of the SCAQS tunnel study with other on-road vehicle emission data. J.Air Waste Manage. Assoc., 40, pp.1495–1504.

Pierson, W.R., A.W. Gertler, N.F. Robinson, J.C. Sagebiel, B. Zielinska, G.A.
Bishop, D.H. Stedman, R.B. Zweidinger and W.D. Ray (1994).
Real-world automotive emissions – summary of studies in the Fort McHenry and Tuscarora mountain tunnels. submitted to Atmospheric Environment.

Post, K. (1979).

Precursor distributions, ozone formation and control strategy options for Sydney. Atmos.Environ., 13, pp.783–790.

Report 2 Emission Registration (1978). Publication of VROM, 's-Gravenhage.

Report 13 Emission Registration (1990). Publication of VROM, ISBN 90 346 2229 0, 's-Gravenhage.

Richter, H.G., F.F. McElroy and V.L. Thompson (1985). Measurement of ambient NMOC concentrations in 22 cities during 1984. Paper No. 85-22.7, presented at the APCA 78<sup>th</sup> Annual Meeting, Detroit, 1985.

Roemer, M., H. Verhagen and H. Diederen (1991). Emissies van koolwaterstoffen, stikstofoxiden en zwaveldioxide in het Rijnmongebied, berekend uit vliegtuigmetingen, en de gevolgen ervan op oxidantvorming. TNO rapport R 91/282, Delft.

Roemer, M.G.M., J.J.M. Berdowski and M.P.J. Pulles (1993). VOC-emissions in the Dutch Rijnmond area as determined by aircraft measurements. Proceedings of the TNO/EURASAPworkshop on the reliability of VOC emission data bases (eds: Baars, Builtjes, Pulles, Veldt), 9–10 June 1993, Delft.

Roemer, M.G.M, P.J.H. Builtjes, P.J. Esser, R.G. Guicherit and T. Thijsse (1995).

 $C_2$ - $C_5$  Hydrocarbon measurements in The Netherlands 1981–1991. Submitted to Atmospheric Environment.

 Roemer, M, P. Builtjes, P. Esser, R. Guicherit, and T. Thijsse (1995).
 C2-C5 hydrocarbon measurements in the Netherlands 1981-1991, P 95/036, TNO MEP, Delft, 1995

Roemer, M. 1995, Emission estimates of  $C_2$ - $C_5$  hydrocarbons in the Rijnmond area on the basis of ground level measurements, TNO report R 95 /203, TNO MEP, Apeldoorn, The Netherlands

Scheff, P.A., Wadden R.A., 1993, Receptor modeling of volatile organic compounds. 1. Emission inventory and validation, Environ. Sci. Technol. 27, 617

#### SDU (1976).

Modellen voor de berekeningen van de verspreiding van luchtverontreiniging inclusief aanbevelingen voor de waarde van parameters in het langetermijnmodel. Staatsuitgeverij, Den Haag.

84 van 87

#### SDU (1981).

Frequentieverdelingen van de luchtverontreinigingsconcentraties; een aanbeveling voor een rekenmethode. Staatsuitgeverij, Den Haag.

#### SDU (1984).

Parameters in het lange-termijnmodel verspreiding luchtverontreiniging; nieuwe aanbevelingen. SCMO-TNO, Delft.

Sedefian, L., S.T. Rao and W.B. Petersen (1981). Editorial Correspondence. Environ.Sci.Technol., 15, pp.364–365.

Sexton, K. (1983).

Evidence of an additive effect for ozone plumes from small cities. Environ.Sci.Technol., 17, pp.402–407.

Sexton, K. and H. Westberg (1979).

Ambient air measurements of petroleum refinery emissions. JAPCA, 29, pp.1149–1152.

Sexton, K. and H. Westberg (1980).

Ambient hydrocarbon and ozone measurements downwind of a large automotive painting plant. Environ.Sci.Technol., 14, pp.329–332.

### Sexton, K. and H. Westberg (1980).

Elevated ozone concentrations measured downwind of the Chicago-Gary urban complex. JAPCA, 30, pp.911–914.

Sexton, K. and H. Westberg (1983).

Photochemical ozone formation from petroleum refinery emissions. Atmos.Environ., 17, pp.467–475.

- Sexton, K. and H. Westberg (1983). Photochemical ozone formation in urban and point-source plumes. Environ.SciTechnol., 17, pp.224–227.
- Sibisi S. and Skilling J. (1996): "Bayesian density estimation." in Maximum Entropy and Bayesian Methods - Cambridge, Engeland 1994 (Kluwer, Dordrecht), p. 189.
- Simpson, D., D.A. Perrin, J.E. Varey and M.L. Williams (1990). Dispersion modelling of nitrogen oxides in the UK. Atmos.Environ., 24A, pp.1713-1733.

<ul> <li>Solberg, S., F. Stordal, N. Schmidbauer and O. Hov (1994).</li> <li>Non-methane hydrocarbons (NMHC) at Birkenes in South Norway, 1988</li> <li>1993. NILU report OR 47/93, ISBN 82-425-0513-6, Kjeller, Noorwegen</li> </ul>	<b>}</b> —
Staehelin, J. and K. Schläpfer (1994). Erfassung von Emissionen des Kraftfahrzeugverkehrs mittels Messungen Gubristtunnel, ETH, Zürich.	im
Stedman, D.H. (1989). Automobile carbon monoxide emission. Environ.Sci.Technol., 23, pp.14 149.	7–
Stedman, D.H. and G.A. Bishop (1990). An analysis of on-road remote sensing as a tool for automobile emissions control. Final report ILENR/RE-AQ-90/05 prepared for the Illinois De- partment of Energy and Natural Resources, Office of Research and Plann	ing.
Stevens, R.K., C.W. Lewis, T.L. Connor, J.F. Collins and R.C. Henry (1993) Receptor modeling and volatile organic compound emissions inventory validation. Proceedings of workshop on VOC assessment and evaluation (eds: Bloemen en Burn), 26–28 January 1993, Amersfoort.	).
Thijsse, T. and W. den Tonkelaar (1987). Meting van de emissies door wegverkeer in de Drechttunnel te Dordrecht TNO rapport R 87/036, Delft.	
Thijsse, T, 1996, personal communication.	
<ul> <li>TNO (1983).</li> <li>Handleiding PLUIMPLUS-Programmapakket, versie 1.1, Hoofdgroep</li> <li>Maatschappelijke Technologie TNO, afdeling Milieuchemie, Delft, 1989</li> </ul>	
Tonkelaar, W den and T. Thijsse (1986). Luchtverontreiniging nabij Moerdijk; evaluatie van 10 jaar onderzoek. Tr rapport R 85/339.	NO
Umweltbehörde (1989). Monatsbericht 4/89 Luftmessnetz Hamburg. Umweltbehörde, Hamburg.	
Vanderstraeten, P.F., E.L. Wauters and G.E. Verduyn (1991). Tunnel air quality; the carbon balance as an alternative to evaluate traffic emissions. Staub, 51, pp.83–90.	C
Van Oss, R. (1997). e-mail communication	

TNO-MEP - R 97/435

86 van 87

Wadden, R.A., I. Uno and S. Wakamatsu (1986). Source discrimination of short-term hydrocarbon samples measured aloft. Environ.Sci.Technol., 20, pp.473–483.

Westberg, H., K. Allwine and E. Robinson (1977).

Ambient hydrocarbon and ozone concentrations near a refinery-Lawrenceville, Illinois 1974. EPA-600/7-77-049.

Westberg, H., K. Sexton and E. Roberts (1981).

Transport of pollutants along the western shore of Lake Michigan. JAPCA, 31, p.385–388.

Westberg, H. and B. Lamb (1985).

Ozone production and transport in the Atlanta, Georgio Region, Report to EPA on Grant No. CR 809221, ASRL, Research Triangle Park, NC, 1985; NTIS Document PB85173839.

#### WVL (1976).

Het Blauwe Boekje: Modellen voor de berekening van de verspreiding van luchtverontreiniging - het lange termijn model, Werkgroep Verspreiding Luchtverontreiniging, Staatsuitgeverij, 's-Gravenhage, 1976

WVL (1981).

Frequentieverdelingen van luchtverontreinigingsconcentraties - een aanbeveling voor een rekenmethode, Werkgroep Verspreiding Luchtverontreiniging,

Staatsuitgeverij, 's-Gravenhage, 1981

Zweidinger, R.B., J.E. Sigsby Jr., S.B. Tejada, F.D. Stump, D.L. Dropkin and W.D. Ray (1988).

Detailed hydrocarbon and aldehyde mobile source emissions from roadway studies. Environ.Sci.Technol., 22, pp.956–962.

## 8. Authentication

Name and address of the contractor:

European Commission DG XI.B2; Gebouw Beaulieu 5; 5/140, Wetstraat 200; 1049 Brussel; Belgium.

Name and function of the co-operators:

M.G.M. Roemer (project leader in Phase I)
R.F. van Oss (Statistical data handling expert)
J.C.Th. Hollander (Air monitoring expert)
Th. R. Thijsse (VOC monitoring expert)
W.A.S. Nijenhuis (Emission Inventory expert)
P.J.M. Valks (Modelling expert)
G.A.C. Boersen (Emission Inventory expert)
H. Verhagen (Air monitoring equipment expert)

Dr. Robert Wilkinson, Mr. Stuart Williamson, Mr. Steve Gillespie, Dr. Matt Reed and Mr. Ian Archibald from Shell Research Ltd. (United Kingdom) are acknowledged for their support to implement the FTIR and HAWK measurements and inputs on the modelling exercise. Mr. Leo Hermans and Mr. Paul Kummu provided excellent support to access the monitoring sites of DCMR in the Rijnmond area.

Names and establishments to which part of the research was put out to contract:

Signature: Approved by: a. Dr. M.P. Keuken Ir. H.P. Baars (Project leader) (Head of Department)

Date upon which, or period in which, the research took place:  $1/2/96\,$  -  $\,1/2/98$ 

TNO-MEP - R 97/435 Annex 1

# Annex 1 Results of the measurements

C2 H2 C2 H4 C2 H6 C3 H6 C3 H8 average number windsector stdev stdev stdev (degrees) of measurements average stdev average average average average stdev 68 0,55 1.89 0.65 4,80 3.49 4.07 1,16 6.87 0.63 4,84 0 6,53 0,77 5.76 5.08 23,29 45 60 0,76 6,47 4,43 21.66 9.54 59 1,43 5.35 2.36 11,92 7,45 7,94 2.45 10.09 2,18 5.81 90 25 1,04 1,56 2,54 3,94 4,34 4,91 1,34 3,09 2,00 4,08 135 6.05 0.75 2.31 5,00 3.69 4.04 1,12 9,06 1,47 180 76 1.44 112 0,85 3,17 2,59 4,55 2,90 3,00 225 0,46 1,93 0.54 0,83 1,55 2,54 77 0,51 0,55 2,75 2,67 2,99 0.60 2.20 0.53 270 0,39 7.01 13,92 0.37 6,42 315 86 0,37 6,50 6,49 3.61 1,47

Table I.1a: Average and standard deviation of air concentrations of C2-C5 at Zwartewaal in ug/m3 at various windsectors.

windsector	average number	C4	H10 (i)	C4	H10 (n)	C5	H12 (i)	C10	H12 (n)
(degrees)	of measurements	average	stdev	average	stdev	average	stdev	average	stdev
0	68	1,24	6,62	0,57	3,05	4,32	11,57	3,39	3,95
45	60	0,74	5,15	0,73	6,18	2,51	8,85	6,98	10,31
90	59	2,66	10,86	1,35	5,61	5,89	15,67	6,91	6,80
135	25	1,42	2,15	1,01	1,73	2,59	3,38	4,12	2,62
180	76	1,13	6,70	0,76	3,37	2,40	6,91	3,66	4,96
225	112	0,57	3,05	0,45	1,89	1,86	4,46	2,54	4,35
270	77	0,59	2,45	0,53	1,54	1,71	3,35	2,72	2,60
315	86	1,43	14,10	0,37	6,49	3,53	17,22	3,62	7,43

hoogvilles

windsector	average number	C2 H2		C2	H4	C2 H6		C3 H6		C3 H8	
(degrees)	of measurements	average	stdev								
0	15		0,3	2,7	0,7	4,7	1,2	2,4	1,2	5,2	1,6
45	9	1,5	0,4	2,5	0,6	3,9	0,8	2,2	0,6	3,5	0,8
90	6	3,1	3,0	3,8	1,5	4,5	0,9	2,3	0,6	4,5	1,5
135	6	2,0	0,4	4,3	0,9	4,0	2,7	2,8	1,0	5,5	1,3
180	4	2,7	1,1	3,6	1,5	3,9	0,4	2,1	0,7	4,3	0,8
225	16	1,6	0,8	2,9	1,5	5,2	1,9	1,7	0,7	4,3	1,8
270	7	1,5	0,7	2,4	1,1	5,8	1,9	2,0	1,1	5,3	2,6
315	12	1,9	1,1	3,5	1,7	8,1	3,8	3,2	1,9	9,0	4,6

Table I.1b: Average and standard deviation of air concentrations of C2-C5 at Hoogvliet in ug/m3 at various windsectors.

windsector	average number	C4	H10 (i)	C4	H10 (n)	C5	H12 (i)
(degrees)	of measurements	average	stdev	average	stdev	average	stdev
0	15	6,53	4,78	20,66	14,67	28,14	24,01
45	9	3,15	1,11	7,87	2,01	10,05	1,84
90	6	6,50	6,00	18,57	15,82	17,20	14,49
135	6	7,42	6,96	22,66	25,96	25,53	29,75
180	4	7,17	4,39	21,75	12,34	24,47	17,81
225	16	4,03	2,17	10,86	6,32	11,97	6,22
270	7	7,13	6,58	18,27	16,25	24,35	28,40
315	12	10,49	7,09	20,61	8,85	21,67	10,62

maassiuis

windsector	average number	C2	H2	C2	H4	C2	H6	C3 H6		C3 H8	
(degrees)	of measurements	average	stdev								
0	17	1,0	0,4	1,9	0,8	3,2	1,5	1,3	0,5	2,9	1,7
45	9	1,2	0,5	2,0	0,8	5,1	4,2	1,7	0,6	3,5	2,3
90	7	1,7	0,7	2,8	1,1	4,1	2,4	1,7	0,5	3,6	1,9
135	3	1,6	0,4	3,3	0,2	3,8	2,1	5,6	4,9	7,2	2,9
180	9	1,5	0,8	2,5	1,6	3,0	2,1	3,5	2,7	4,5	2,6
225	16	1,8	1,1	2,8	1,4	5,0	4,0	1,9	0,7	6,8	2,7
270	8	1,2	0,6	2,0	0,6	3,9	1,3	1,7	1,0	5,4	2,6
315	14	1,4	0,8	2,2	1,1	4,1	2,1	1,6	0,7	5,8	5,6

Table I.1c: Average and standard deviation of air concentrations of C2-C5 at Maassluis in ug/m3 at various windsectors.

windsector	average number	C4 H10 (i)		C4	H10 (n)	C5	H12 (i)
(degrees)	of measurements	average	stdev	average	stdev	average	stdev
0	17	2,59	2,00	7,25	6,14	10,84	6,85
45	9	1,97	1,99	5,19	4,18	8,63	5,11
90	7	2,64	2,62	8,88	9,99	14,26	11,66
135	3	3,81	0,37	9,51	1,32	13,98	2,09
180	9	3,60	2,66	10,17	9,48	13,71	12,45
225	16	5,63	2,79	16,16	8,65	21,27	11,15
270	8	3,67	3,27	11,80	12,31	16,39	14,18
315	14	2,86	1,77	8,33	4,71	11,17	6,67

windsector	average number	benzene		toluene		ethyl	benzene	m-xylene	
(degrees)	of measurements	average	stdev	average	stdev	average	stdev	average	stdev
0	236	4,1	3,6	9,1	6,0	2,3	1,4	4,0	2,7
45	111	3,2	2,9	7,9	6,7	2,5	2,1	4,3	4,0
90	35	3,2	3,4	8,0	2,3	2,3	2,0	3,9	3,8
135	32	2,4	1,4	6,1	4,0	1,7	1,1	2,8	2,2
180	115	2,7	2,2	7,2	6,4	2,0	1,6	3,0	2,8
225	164	2,1	2,5	4,7	3,6	1,3	1,0	2,0	1,7
270	226	2,6	2,1	5,0	3,6	1,4	1,1	2,1	1,8
315	236	4,2	3,3	6,1	3,9	1,8	1,1	2,6	1,6

Table I.2a: Average and standard deviation of air concentrations of BTX at Hoogvliet in ug/m3 at various windsectors.

windsector	average number	p-xylene		o-xylene		styrene	
(degrees)	of measurements	average	stdev	average	stdev	average	stdev
0	236	1,7	1,7	2,2	1,4	0,9	2,3
45	111	2,4	1,4	2,4	2,1	4,0	6,1
90	35	2,1	2,0	2,2	2,1	0,4	0,6
135	32	1,2	1,2	1,6	1,2	0,4	0,3
180	115	1,5	1,4	1,7	1,5	0,4	0,4
225	164	0,9	0,9	1,2	1,0	0,4	0,3
270	226	1,0	1,0	1,3	1,1	0,7	1,1
315	236	1,4	1,0	1,5	0,9	0,6	0,8

windsector	average number	benzene		toluene		ethyl	benzene	m-xylene	
(degrees)	of measurements	average	stdev	average	stdev	average	stdev	average	stdev
0	502	2,0	1,9	4,5	4,4	2,1	3,8	2,6	2,9
45	279	1,8	1,3	4,2	3,1	2,0	1,3	2,3	1,9
90	118	2,4	1,4	6,6	4,3	2,4	1,3	3,8	2,7
135	51	4,3	3,4	8,6	5,9	3,0	1,6	5,0	4,0
180	269	3,7	2,2	6,2	4,5	3,4	2,5	3,4	2,8
225	264	3,3	3,8	4,8	4,1	2,1	1,6	2,7	2,8
270	250	2,4	6,0	4,1	3,3	1,7	0,7	2,3	1,8
315	315	1,5	0,9	3,4	2,8	1,6	0,8	2,1	1,5

Table I.2b: Average and standard deviation of air concentrations of BTX at Maassluis in ug/m3 at various windsectors.

windsector	average number	p-xylene		o-xylene		styrene	
(degrees)	of measurements	average	stdev	average	stdev	average	stdev
0	502	1,1	1,3	1,4	1,4	0,8	0,6
45	279	0,9	0,7	1,2	0,9	0,5	0,3
90	118	1,4	1,0	1,9	1,3	0,6	0,4
135	51	1,9	1,3	2,3	1,6	0,6	0,3
180	269	2,3	1,6	1,9	1,9	0,7	0,7
225	264	1,9	1,9	1,5	1,6	0,6	0,4
270	250	1,5	1,6	1,2	0,9	0,6	0,4
315	315	0,8	0,6	1,1	0,8	0,5	0,3

sampler	benzene		toluene		ethylbenze	ne	p+m-xylen	e	o-xylene		styrene	
code	average	stdev	average	stdev	average	stdev	average	stdev	average	stdev	average	stdev
P1	3,1	0,9	3,8	1,1	1,3	0,8	3,0	1,3	0,8	0,3	0,0	0,1
P2	5,4	2,5	6,7	2,7	2,2	0,9	8,6	2,8	1,8	0,9	0,5	0,2
P3	4,5	2,2	6,4	2,5	1,6	0,9	6,6	5,5	2,2	1,4	0,2	0,3
P4	7,6	5,2	9,2	3,4	3,9	1,7	6,8	2,6	2,6	1,5	3,3	2,3
P5	13,0	8,5	21,6	8,4	2,6	0,9	8,2	2,4	2,6	0,7	0,4	0,4
P6	18,8	16,2	16,0	3,5	4,8	1,6	9,1	1,1	3,7	1,2	2,4	0,9
P7	9,8	5,6	10,6	1,2	3,6	1,1	7,1	1,3	2,6	0,9	1,1	0,5
P8	4,1	1,8	10,1	8,8	1,8	0,5	3,9	1,1	1,7	0,4	0,4	0,4
P9	3,9	0,9	10,0	2,9	1,5	0,3	3,2	0,8	1,2	0,4	0,4	0,3
P10	4,4	1,8	10,9	4,5	2,4	0,9	5,8	2,8	2,3	0,9	1,3	0,6
P11	7,9	2,6	20,3	4,6	3,9	1,2	9,6	2,9	3,7	0,9	1,2	0,4
P12	8,0	6,0	15,7	10,2	3,2	1,8	7,8	4,3	3,0	1,3	1,0	0,6
P13	3,8	1,3	9,1	3,5	2,3	0,6	5,6	1,5	2,1	0,7	2,1	0,9
P14	3,1	0,8	7,0	1,6	2,2	0,4	4,7	0,8	2,0	0,4	12,6	9,0

Table II.3A: Average and standard deviation of VOC air concentrations in ug/m3 measured during six periods in the Botlek/Pernis area with passive samplers.

sampler	124-trimeth	nylbenzene	n-decane		n-undecan	е	cyclohexar	ne	methylcyclohexane		hexane	
code	average	stdev	average	stdev	average	stdev	average	stdev	average	stdev	average	stdev
P1	0,5	0,4	0,4	0,4	0,4	0,3	3,6	2,4	3,1	4,1	2,3	1,4
P2	1,3	0,8	0,7	0,2	0,8	0,5	4,0	2,5	1,5	0,9	4,6	2,5
P3	0,8	0,2	0,5	0,2	0,5	0,4	2,9	2,8	1,6	0,5	2,9	1,0
P4	1,4	0,4	1,7	0,9	1,5	0,7	20,3	20,6	2,9	0,9	5,8	1,9
P5	1,1	0,3	0,6	0,3	0,5	0,4	10,2	10,7	2,5	0,8	4,9	1,3
P6	2,6	0,4	2,0	0,4	1,8	0,4	4,9	1,6	4,6	1,3	11,3	2,6
P7	1,7	0,2	1,3	0,3	1,4	0,3	2,9	1,5	2,8	0,8	7,3	1,5
P8	1,6	0,5	1,1	0,5	1,4	0,6	2,1	0,5	2,1	0,6	8,2	5,0
P9	1,2	0,4	0,9	0,3	1,3	0,4	1,5	0,3	1,8	0,4	6,0	1,6
P10	1,6	0,5	1,0	0,4	1,0	0,3	1,6	0,4	2,5	0,8	7,7	2,5
P11	2,4	0,7	1,6	0,7	1,7	0,6	1,8	0,3	3,8	1,1	10,8	3,0
P12	1,9	0,7	1,2	0,4	1,3	0,4	2,0	1,0	6,0	4,4	12,0	6,3
P13	2,1	0,5	2,4	1,4	5,0	3,9	1,8	0,3	2,7	0,8	7,4	2,6
P14	1,4	0,3	0,9	0,1	1,1	0,3	2,0	1,0	2,3	1,0	6,4	3,7

TableII.3b: Air concentrations in ug/m3 during three periods at Zwartewaal, Maassluis and Hoogvliet measured with passive samplers.

location	period	benzene	toluene	ethyl-	p+m-	0-	styrene	n-decane	n-un-	cyclo-	methyl-	hexane
				benzene	xylene	xylene			decane	hexane	cyclohex.	
	16/4-14/5											
Zwartewaal	14/5-10/6	1,2	3,4	0,5	1,4	0,6	0,0	0,4	0,3	0,8	0,7	2,2
	10/6-9/7	0,6	1,7	0,3	0,6	0,2	0,0	0,3	0,0	0,6	0,4	1,3
Maassluis	16/4-14/5		3,7	1,0	4,5	0,7	0,3	0,3	0,4	2,7	1,8	5,6
	14/5-10/6	1,8	4,7	0,9	2,7	1,1	0,0	0,3	0,3	0,9	1,0	3,2
	10/6-9/7	1,3	3,0	0,6	1,6	0,7	0,0	0,0	0,0	0,9	0,7	2,3
Hoogvliet	16/4-14/5	3,6	7,3	1,3	3,8	1,4	0,5	0,5	0,5	1,0	1,5	4,8
	14/5-10/6	2,5	5,9	1,2	3,1	1,2	0,7	0,5	0,4	1,1	1,3	4,5
	10/6-9/7	2,5	5,5	1,0	2,7	1,0	0,4	0,3	0,4	1,1	1,1	3,8

Table II.4: Average and standard deviation of VOC air concentrations in ug/m3 in Hoogvliet, Maassluis and Zwartewaal measured by tenax sampling tubes from 10 AM till 16 PM in the period mid April till mid July.

component	Hoogvliet (n=69)		Maassluis	(n=70)	Zwartewa	(n=57)
	average	st.dev.	average	st.dev.	average	st.dev.
hexane	3,6	5,4	4,9	5,0	3,6	4,1
trichloromethane	0,1	0,0	0,1	0,0	0,0	0,0
methylcyclopentane	1,4	2,3	1,8	1,7	0,8	1,1
2,4-dimethylpentane	0,3	0,4	0,3	0,3	0,1	0,1
1,1,1-trichloroethane	0,2	0,2	1,5	4,4	0,3	0,3
benzene	2,5	2,8	2,5	2,5	1,4	1,9
tetrachloromethane	0,5	0,5	0,5	0,4	0,5	0,3
2-methylhexane	1,4	1,9	1,3	1,1	0,5	0,7
3-methylhexane	1,0	1,3	0,9	0,8	0,3	0,4
2,2,4-trimethylpentane	1,2	1,4	1,0	1,0	0,3	0,5
trichloro-ethene	0,1	0,1	0,3	0,6	0,1	0,1
n-heptane	1,6	2,0	1,8	1,7	1,0	1,0
methylcyclohexane	0,6	0,9	1,0	1,3	0,3	0,6
2,5-dimethylhexane	0,2	0,2	0,2	0,2	0,0	0,0
2,4-dimethylhexane	0,3	0,3	0,3	0,3	0,1	0,1
toluene	10,2	14,3	6,2	6,6	2,8	4,7
3-methylheptane	0,3	0,4	0,3	0,3	0,1	0,1
n-oktane	0,6	0,7	0,7	0,6	0,5	0,8
tetrachloorethene	0,3	0,3	0,1	0,2	0,1	0,1
ethylbenzene	1,7	2,1	1,7	1,6	0,6	0,7
p,m-xylene	4,5	5,5	4,0	4,1	1,2	1,2
styrene	0,6	1,0	0,2	0,2	0,1	0,2
o-xylene	1,8	2,1	1,8	1,6	0,5	0,5
n-nonane	1,0	1,0	1,2	1,1	1,0	0,9
iso-propylbenzene	0,2	0,3	0,1	0,2	0,1	0,0
n-propylbenzene	0,4	0,4	0,4	0,3	0,1	0,1
3-ethyltoluene	1,7	1,9	1,4	1,1	0,4	0,4
4-ethyltoluene	0,6	0,7	0,6	0,5	0,1	0,1
1,3,5-trimethylbenzene	0,3	0,4	0,3	0,3	0,1	0,1
2-ethyltoluene	0,4	0,5	0,4	0,4	0,1	0,2
1,2,4-trimethylbenzene	1,8	2,4	1,5	1,3	0,3	0,4
n-decane	0,9	1,1	1,0	1,0	0,8	1,1
1,2,3-trimethylbenzene	0,3	0,3	0,3	0,4	0,1	0,1
n-undecane	0,7	0,8	0,7	1,1	0,4	0,4
naftalene	0,2	0,3	0,1	0,2	0,1	0,0
n-dodecane	0,3	0,4	0,2	0,3	0,4	0,7

Pagina 1